

A Service of

ZBW

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Ornaghi, Carmine; Van Beveren, Ilke; Vanormelingen, Stijn

Working Paper The impact of service and goods offshoring on employment: Firm-level evidence

NBB Working Paper, No. 319

Provided in Cooperation with: National Bank of Belgium, Brussels

Suggested Citation: Ornaghi, Carmine; Van Beveren, Ilke; Vanormelingen, Stijn (2017) : The impact of service and goods offshoring on employment: Firm-level evidence, NBB Working Paper, No. 319, National Bank of Belgium, Brussels

This Version is available at: https://hdl.handle.net/10419/173775

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



WWW.ECONSTOR.EU

The impact of service and goods offshoring on employment : Firm-level evidence



by Carmine Ornaghi, Ilke Van Beveren and Stijn Vanormelingen

May 2017 No 319



Editor Jan Smets, Governor of the National Bank of Belgium

Statement of purpose:

The purpose of these working papers is to promote the circulation of research results (Research Series) and analytical studies (Documents Series) made within the National Bank of Belgium or presented by external economists in seminars, conferences and conventions organised by the Bank. The aim is therefore to provide a platform for discussion. The opinions expressed are strictly those of the authors and do not necessarily reflect the views of the National Bank of Belgium.

Orders

For orders and information on subscriptions and reductions: National Bank of Belgium, Documentation - Publications service, boulevard de Berlaimont 14, 1000 Brussels

Tel +32 2 221 20 33 - Fax +32 2 21 30 42

The Working Papers are available on the website of the Bank: http://www.nbb.be

© National Bank of Belgium, Brussels

All rights reserved. Reproduction for educational and non-commercial purposes is permitted provided that the source is acknowledged.

ISSN: 1375-680X (print) ISSN: 1784-2476 (online)

Abstract

Advances in communication technology have led to a remarkable increase in the tradability of services, resulting in a substantial increase in offshoring of services over the last two decades. Research investigating how this surge in service offshoring affects employment, has been largely hampered by the paucity of suitable microdata. This paper tries to fill this gap by using a newly constructed database of Belgian firms that combines individual transaction-level data on international trade in goods and services with annual financial accounts. This unusually rich dataset allows us to produce fresh evidence on the impact of goods and service offshoring industries and the service sectors. Our results show that: (i) goods offshoring has a positive impact on employment growth among workers with both low and high levels of education in the manufacturing industry but this effect disappears when controlling for scale effects; and (ii) service offshoring has a negative impact on employment growth among highly educated workers in the service sectors. This novel evidence suggests that globalization may threaten job security of higher educated workers too.

Keywords: Offshoring, trade in services, labour markets

Authors

Carmine Ornaghi, University of Southampton, e-mail: <u>C.Ornaghi@soton.ac.uk</u> Ilke Van Beveren, CBS The Netherlands and KU Leuven, e-mail: <u>ilke.vanbeveren@kuleuven.be</u> Stijn vanormelingen, KU Leuven, e-mail: <u>Stijn.vanormelingen@kuleuven.be</u>

Acknowledgements

We would like to thank Sebastien Miroudot, Hector Calvo Pardo, Jan Van Hove, an anonymous referee of the National Bank of Belgium and participants at the CEP Workshop on Employment Effects of Service Trade Reform in Geneva, the EIST workshop in Florence and the ETSG conference in Paris for helpful comments. This paper has also benefited from a presentation at the OECD. This paper was initiated while llke Van Beveren was visiting the National Bank of Belgium.

The views expressed in this paper are those of the authors and do not necessarily reflect the views of the National Bank of Belgium or any other institution to which the authors are affiliated. Responsibility for the information and views expressed in the paper lies entirely with the authors.

TABLE OF CONTENTS

Ι.	Introduction	1
II.	Data	5
III.	Empirical Framework and Instruments	.10
IV.	Results	.15
	A. Main Results	.16
	B. Robustness Checks and Extensions	.21
۷.	Conclusion	.23
Refer	ences	.26
Figure	es	. 30
Table	S	. 32
Appe	ndices	.41
Natio	nal Bank of Belgium - Working papers series	.51

I. Introduction

The impact of globalisation on domestic labour markets has been debated among academics for several years and, recently, has taken centre stage in the political debate in various developed countries, most recently in the 2016 US election campaign. The relocation of jobs abroad by domestically active firms is generally considered one of the culprits for the decline in employment in developed economies' manufacturing industries as well as for the worse labour market outcomes of unskilled workers. Given that unskilled jobs are more easily moved offshore to developing countries that have a comparative advantage in the production of unskilled labour-intensive goods, globalisation can lower the relative demand for unskilled workers in developed countries, thereby contributing to increased wage inequality in Western economies. While trade was originally found to explain only a small part of the observed changes in wage inequality, the introduction of offshoring – the disintegration of the production process across national borders – in theoretical and empirical work has shown that globalisation accounts for a substantial part of the rise in wage inequality (Krugman, 2008; Feenstra and Hanson, 2003).

Because of continuous advances in information and communication technology, services have also become increasingly tradable. Although the level of international trade in services is still relatively small, the growth in services trade substantially outperforms the growth in goods trade. This evolution has led governments to start new trade negotiations that can reflect the reality of increasingly service-based international commerce; for instance, several members of the WTO are currently negotiating the Trade in Services Agreement (TiSA) with the aim of facilitating market access and improving rules in areas such as licensing, telecommunications, e-commerce and financial services. While most economists believe that opening up markets for services leads to economic growth, increasing tradability also means that a large number of jobs previously insulated from foreign competition may potentially be located abroad (Blinder, 2006), thus leading to what Amiti and Wei (2005) call the "fear of service offshoring."

When a firm relocates part of its production process abroad, one would expect a direct negative effect on firm employment, the magnitude of which depends on the number of in-house workers that are attached to the offshored process. However, offshoring may also lead to productivity gains and product quality improvements that can boost firms' output growth and employment (Wright, 2014). Because the question of which of the two effects dominates is an empirical one, the aim of this paper is to produce fresh empirical evidence on the impact of offshoring on employment outcomes. More precisely, we use a rich dataset of Belgian firms to investigate the effects of both goods and service offshoring on total labour demand, as well as on relative labour demand for skilled and unskilled workers in the manufacturing and services sectors.

Most of the academic literature on the labour market effects of trade has focused on the relative demand for unskilled labour, while studies on the absolute change in employment are less common. This may reflect modelling conventions that impose inelastic labour supply and abstract from labour market frictions. As a result, workers losing their jobs in firms/sectors because of increased globalisation would simply be relocated to other firms/sectors (Acemoglu et al., 2014). However, with (labour) market imperfections, there is no guarantee that such perfect relocation would occur. For example, Autor et al. (2013); Acemoglu et al. (2014) and Pierce and Schott (2016) find that import competition from China has been a major force behind employment reductions in the US manufacturing sector. Moreover, the import competition has depressed overall job growth in the US, indicating imperfect relocation.

Most of the studies on the impact of offshoring on labour have focused on offshoring of goods, initially using industry-level data (see, for example, Feenstra and Hanson, 1999) and more recently using firm-level data (e.g., Mion and Zhu, 2013). The general consensus in this literature is that the relative demand for unskilled workers falls in response to goods offshoring. However, the findings from studies investigating the effects on absolute employment are mixed, with most of them estimating rather small effects of offshoring on domestic employment, whether positive or negative (Wright, 2014). In a recent article, Monarch et al. (2017) study several offshoring events by matching Trade Adjustment Assistance (TAA) program petition data to U.S. Census Bureau microdata. They find that offshoring firms experience a significant decline in employment in the year when offshoring starts and in the following 4 years after the event. They also show that there is no evidence of employment recovery in the longer run.

Compared to the large empirical and theoretical literature on trade in goods, our understanding of trade in services and its impact is still hampered by the paucity of suitable microdata. Given that service tasks are, on average, more skill-intensive than production tasks, economic intuition would suggest that service offshoring should exert a downward pressure on skilled labour demand (Crino, 2009). But the few available empirical studies have found no support for this hypothesis. On the contrary, service offshoring seems to increase relative demand for skilled workers (Crino, 2010b,a, 2012), mimicking the findings for goods offshoring in earlier studies. But Crino (2012) shows that this similarity between service and goods is actually driven by two different forces: complementarity between imported services and domestic skills in the case of service offshoring and substitution of low-skilled labour in the case of material offshoring. Similarly, Geishecker and Görg (2013) combine individual-level data on wages with sectorlevel data on service offshoring and find that service offshoring negatively affects the real wage of low-and medium-skilled individuals in the same industry, while skilled workers benefit from service offshoring through higher real wages. Using a panel of Swedish manufacturing firms, Andersson et al. (2016) also finds that service offshoring increases relative demand for skilled labour while goods offshoring appears to have no impact.

The number of studies that investigate the effective impact of service offshoring on total labour demand is even more limited. Amiti and Wei (2005, 2006) use sector-level data for UK and US manufacturing, respectively. They find a small negative or no effect of service offshoring on labour demand in the US, but a small positive effect in the UK. Görg and Hanley (2005) uses a panel of 100 Irish electronic firms and find some negative effects of service offshoring on employment, but smaller in magnitude than those of goods offshoring. Using data on imports and exports of service retrieved from a survey of UK firms, Hijzen et al. (2011) find that employment growth in firms that import services, is higher than in firms that do not. The results appear to be driven by cost savings associated with offshoring that give rise to an increase in the scale of production.¹

We aim to contribute to the existing literature in several ways. First, we investigate the relationship between employment outcomes and the offshoring of both goods and services. The number of studies measuring the effects of service offshoring is still limited, and they use mostly aggregated sector-level data or are based on surveys of a sample of firms. In contrast, we can observe firm-level imports of both goods and services. Second, we incorporate the services sectors in our analysis. Although the bulk of employment is located in the services sectors, most of the empirical work focuses solely on manufacturing industries, largely because of data constraints. This is particularly important for our research question, as most of the activity in service offshoring takes place in the services sectors. Third, we look at both the impact on the change in total employment and the effect on workers with different education levels. Finally, our econometric framework will allow us to disentangle substitution and scale effects, as well as short-run effects versus the impact over a longer period. In principle, purchasing an input from abroad implies the immediate replacement of a task previously done by domestic workers employed either by the importing firms or by a domestic supplier of that firm. However, the purchase of foreign inputs may

¹Another strand of the literature has investigated the impact of globalisation decisions on firm performance. Amiti and Konings (2007) show how the imports of intermediate products lead to higher productivity. Similarly, Halpern et al. (2011) provide evidence that the access to new intermediate goods varieties is an important source of productivity growth. Goldberg et al. (2010) find that access to foreign imports increases firm product scope and that this can lead to an increase in the market shares of offshoring firms, thus mitigating or even offsetting the initial negative effect of offshoring on employment.

later lead to productivity gains and quality improvements that allow the firm to expand output and employment (Grossman and Rossi-Hansberg, 2008).

Our findings indicate that goods and service offshoring have different effects on employment growth. Employment growth is positively related to goods offshoring in the manufacturing sector. However, when controlling for scale effects, this relation becomes either non-statistically different from zero or even negative, depending on the specification. Service offshoring is negatively related to employment growth in the services sector, and interestingly enough, this result appears to be driven by employees with high education. To the best of our knowledge, this is the first paper to find robust large-scale evidence that in the services sectors, it is the job security of skilled workers that globalisation threatens. This finding is compatible with the fact that most of service imports come from other OECD countries.

The rest of the paper is structured as follows. Section II describes the dataset used for the analysis. Section III explains the empirical methodology and the construction of the instrumental variables used to identify the causal impact of offshoring on employment. Section IV presents the results, and section V concludes.

II. Data

For our empirical analysis, we rely on Belgian firms' annual accounts which are merged with data on their trading activities for the period 1996-2005.² The annual accounts contain the standard variables, such as employment, value added, turnover and book value of tangible assets.³ While all limited-liability firms

²We restrict the analysis to the period 1996-2005 for two reasons. First, the definition of employment in the annual accounts has changed in 1996. Since many firms still reported employment according to the old reporting standard, the data for 1995 are not always reliable, hence we drop this year from the sample. Second, trade in services is available for the full population only until 2005. Until 2005, the data were constructed using transaction data on cross-border payments to foreign enterprises reported by financial institutions, implying that data were available for the full population of firms. Starting in 2006, data are collected directly from a sample of (large) firms, resulting in a structural break in the series. Therefore, we include only data until 2005 in our sample.

 $^{^{3}}$ For a limited number of firms, the reporting year does not run from January to December. To correct for these, we follow a procedure suggested by the National Bank of Belgium to annualize these

are required to report their annual accounts to the National Bank of Belgium, reporting requirements are limited for small firms.⁴ Since some of the variables used in the empirical analysis, such as turnover and job flows by education levels, are available only for firms that report complete annual accounts, we limit our attention to these (larger) firms.

We obtained data on trade in goods from the National Bank of Belgium (NBB). We observe for each firm in a given year its imports and exports at the productcountry level, where products are defined according to the Combined Nomenclature (CN8) classification. Data on intra-EU imports and exports of goods are subject to reporting cutoffs that have changed over time. Specifically, from 1995 to 1997, firms had to report intra-EU exports (imports) if their value in the previous year exceeded $\leq 104,115$. Between 1998 to 2005, this threshold increased to $\leq 250,000$ per year. Moreover, the change in EU membership in 2004 implicitly increased the number of countries included in intra-EU declarations. For extra-EU trade flows, data are collected from customs data. All transactions for which the value is higher than $\leq 1,000$ or whose weight is over 1,000Kg have to be recorded.

The international trade in services data for Belgium are collected on the basis of the balance of payment returns sent to the NBB either by commercial banks or by direct reporters.⁵ Between 1995 and 2005, banks had to report payments made or received by their clients when the payer or the payee was a non-resident. Data are available at the firm-service-country-year level. Starting with the NACE rev1.1

accounts.

⁴Firms are considered small if they do not exceed more than one of three criteria: 1)employment higher than 50; 2) balance sheet total exceeding €3.65 million; and 3)turnover higher than €7.3 million. ⁵Trade in services is usually classified in four different modes (Francois and Hoekman, 2010). Mode *1* is cross-border supply and applies when service suppliers located in one country provide services in another country without either the buyer or the supplier moving to the physical location of the other. A typical example is a call center located in India that provides services to a Belgian firm. Mode 2 is consumption abroad and applies when the service is consumed by a resident of one country in the territory of another country–e.g., hotel services to tourists. Mode 3 refers to commercial presence–i.e. firms moving to the location of the consumers to provide their services locally through the establishment of a foreign affiliate or branch. Mode 4 is the movement of natural persons and refers to services provided by the firm of one country through the presence of natural persons in another country–for example, a Polish transport company that offers transport services in Belgium through the presence of its trucker in Belgium. Since the data for Belgium are constructed using financial transaction data involving foreign businesses and Belgian residents, mode 3 is not included, as there is no cross-border payment involved in this mode of service trade.

classification used by the NBB, we divide services into seven different groups: Transport, Financial and Insurance, Communication, Information Technologies, (other) Business, Construction and Cultural Services.⁶ In our empirical analysis, we exclude construction and cultural services.

In constructing our dataset, we limit attention to transactions involving Belgian firms, thus excluding for example spending by Belgian tourists abroad. We also exclude the following transactions: 1) transactions in goods that did not involve a change in ownership; 2) payments of royalties and license fees and financial flows between related companies since these could reflect profit shifting for tax reasons; 3) all merchanting transactions related to trade in goods and 4) payments made by governments or international institutions.

Figure 1 shows the evolution of trade in goods and trade in services in nominal values. The growth in trade of services is higher than that for goods over the time window considered. As shown in Appendix VII.B, this increase is even more pronounced for services such as Business Services, (+254% increase in nominal value of imports), IT Services (+271%) and Communication Services (+340%), compared to an average of +110% for trade in goods. However, in level terms, imports of services still represent only around 10% of the value of imports of goods at the end of our sample period.⁷

To convert the data into real values, we use deflators from the EUKLEMS database. The international trade data – both services and goods – are deflated with the output deflator corresponding to the NACE code of the trade flow.⁸ Firms' turnover, material inputs and value added reported in the annual account dataset are deflated with the deflators for output, material and value added, respectively. To convert tangible and intangible fixed assets into real values, we

 $^{^6{\}rm The}$ NACE rev1.1 classification translates one-to-one into the international classification of services EBOPS. Table A.1 in Appendix VII.A shows the EBOPS classification and the corresponding NACE codes.

codes. $$^7{\rm The\ coverage}$$ of the international trade in services is much smaller as banks are not included in the annual report database.

 $^{^{8}}$ For trade in goods, CN8 products are translated into NACE categories using an appropriate correspondence table. Manufacturing firms are firms active in NACE rev 1.1 sectors 15 to 36. The services sector consists of NACE rev 1.1 codes 60 to 74.

apply the capital deflator reported by Eurostat for Belgium in the appropriate year. Finally, we use the Harmonized Consumer Price Index (HCPI) to deflate wages.

After some cleaning of the data,⁹ the final sample that we use in our empirical analysis consists of an unbalanced panel of 3751 manufacturing firms and 3679 service firms for the period 1996 to 2005. Table 1 displays total imports of the different types of services for this final sample.¹⁰ Services imports have increased by more than imports of goods, mostly due to the boom in business, IT and communication services. Although the sample consists of large firms, only a minority imports services, but the share of firms importing services has increased from 28.2% to 38.8% over the sample period. The share of firms that imports goods is substantially higher, at more than 50% (and over 80% if we consider only firms in the manufacturing sector). These large numbers reflect the Belgium's substantial trade openness as well as the fact that we focus on large firms.

We define offshoring of goods and services as the firm-level imports of goods and services, respectively. In the empirical section we check the robustness of our results using the following alternative definitions of offshoring: (a) "narrow goods offshoring" defined as the purchases of inputs classified in the same NACE2 manufacturing industry of the firm and, similarly, "core service offshoring" defined as acquired services that belong to the same group classification discussed previously (see last column of Table A.1 in the Appendix); and (b) offshoring from OECD and non-OECD countries.

In Figure 2, we plot the average ratio between firm-level offshoring and total expenditures on material and service inputs for each NACE 2-digit sector. Offshoring of goods and of services are shown in light-grey and dark-grey, re-

 $^{^{9}}$ We drop observations where (i) the share of offshoring–i.e., the ratio between offshoring of goods and services and turnover, is above one; and (ii) the absolute change in the share of offshoring over two consecutive years is above 0.5. Furthermore, we winsorize the growth of employment and share of offshoring at the 1st and 99th percentile.

¹⁰The relative importance of each service category is comparable to the full dataset. Only the Financial and Insurance Services imports are of relatively lower importance in comparison to the figures in Table B.1, which show the importance of the different types of services in the trade database. This is because banks are not included in the firm-level dataset.

spectively. Several sectors in the manufacturing compartment are found to import more than 20% of their intermediary inputs. Not surprisingly, offshoring of intermediate inputs is mostly prevalent in the manufacturing industries, while offshoring of service inputs is more common in service sectors. Total offshoring is higher in manufacturing industries, reflecting higher tradability of goods in comparison with services.

Table 2 reports summary statistics for some of the key variables used in the empirical analysis. The average firm in the manufacturing sector employs 153 people and generates 45 million euro in revenues. The share of offshoring of goods, defined as the ratio between offshoring and turnover, is almost 20%, while the corresponding share for services is around 0.5%. Average employment growth is very low and equal to 0.45%. The average firm active in the services sector employs around 156 people, with a turnover of almost 22 million euro. In general, total imports as a percentage of turnover is relatively low compared to the manufacturing sector, but imports of services are much more important and equal to 3.3%. Average employment growth is equal to 3.6%, reflecting the aggregate pattern of increasing importance of the services sector.

GROWTH OF EDUCATION. — Our data also report the inflow and outflow of employees (but not their total stock) by education level. The employees are divided into 4 categories: (1) primary education or lower; (2) secondary education; (3) higher education, non-university; and (4) university education. We collapse these four categories into two education levels: low education (categories 1 and 2) and high education (categories 3 and 4). To gain insight into the relative importance of workers with low (L) and high (H) education in the manufacturing and services sectors, we compute the share of group j=L, H in the total gross flow of employees as $Share^{j} = \frac{Inflow^{j}+Outflow^{j}}{\sum_{j=L,H} (Inflow^{j}+Outflow^{j})}$, and we find that employees with high education account for 40% of total gross flows in the service sector, compared to a much lower 20% in the manufacturing sector.

To compute the growth rate of employment by education level, we use the following procedure. First, we use the observed net flow of employment $\Delta N_{it}^{j} = Inflow^{j} - Outflow^{j}$ in education group j to compute the corresponding net flows in total employment, $\Delta N_{it} = \sum_{j=L,H} \Delta N_{it}^{j}$. Second, we compute the difference between ΔN_{it} and the change in total average number of employees ΔE_{it} ,¹¹ and we add this difference to the net flows of low-education or high-education workers proportional to their share in the sum of the absolute values of these net flows. For example, suppose that ΔE is equal to 2, ΔN^{L} equal to -8 and ΔN^{H} equal to 4, resulting in ΔN being equal to -4. The difference between ΔE and ΔN is, thus, equal to 6. The share of low-(high-) skilled workers in the total flow is $\frac{8}{8+4} = \frac{2}{3}$ (respectively, $\frac{1}{3}$). The adjusted flows are then $\Delta E^{L} = -8 + \frac{2}{3}6 = -4$ and $\Delta E^{H} = 4 + \frac{1}{3}6 = 6$. Finally, we use the adjusted flows ΔE^{j} in each education group to compute the growth rate of employment by education levels as $\Delta e_{it}^{j} = \frac{\Delta E_{it}^{j}}{0.5(E_{it-1}+E_{it})}$.¹²

III. Empirical Framework and Instruments

A. Framework

The specification we take to the data consists in regressing the growth of employment, Δe_{it} , on time-varying firm-level measures of share of offshoring for goods, SO_{it}^g , and services, SO_{it}^s , measured as imported goods or imported services over total turnover.¹³ We define the dependent variable as growth rates because, as explained above, we observe changes in employment (but not employment levels) for different education levels. We use measures of offshoring scaled by firm size to eliminate the possibility of finding a positive (or negative) correlation be-

¹¹The difference between ΔN and ΔE may be due to two reasons. First, ΔN refers to the difference between employment at the end of year t and employment at the end of year t-1, while ΔE refers to the change in average employment during the year. Second, there could be measurement error in reporting the flows by education levels.

¹²Following the notation just introduced, in the rest of the paper, we use Δx to indicate the percentage change (i.e. growth rate) of variables X and ΔX to indicate the difference between the levels of X over two consecutive years.

 $^{^{13}}$ See also Bernard et al. (2006) and Mion and Zhu (2013) for similar approaches

tween imports and employment due to exogenous positive (respectively, negative) shocks in demand or productivity that cause a proportional increase (decrease) in both output and variable inputs: for instance, the award of a public procurement contract that leads to a simultaneous increase in output and imported inputs but that leaves the share of offshoring unchanged.

To capture differences between the immediate impact of offshoring and the effects over a longer period, we start with the following specification:

(1)
$$e_{it} = \beta_1 e_{it-1} + \beta_2 SO_{it}^g + \beta_3 SO_{it-1}^g + \beta_4 SO_{it}^s + \beta_5 SO_{it-1}^s + \eta_i + \varepsilon_{it},$$

where η_i is the firm unobserved productivity assumed to be constant over the time window considered and ε_{it} captures shocks to employment, possibly correlated with the offshoring activities, and measurement errors. Subtracting e_{it-1} on both sides of the equation and subtracting and adding $\beta_2 SO_{it-1}^g$ and $\beta_4 SO_{it-1}^s$ on the RHS, we obtain:

(2)
$$\Delta e_{it} = (\beta_1 - 1)e_{it-1} + \beta_2 \Delta SO_{it}^g + (\beta_2 + \beta_3)SO_{it-1}^g + \beta_4 \Delta SO_{it}^s + (\beta_4 + \beta_5)SO_{it-1}^s + \eta_i + \varepsilon_{it},$$

where ΔSO_{it}^g (ΔSO_{it}^s) indicates the difference in the share of goods (respectively, service) offshoring over two consecutive years. The coefficients on ΔSO_{it} measure the short-run effect of goods or service offshoring, while the coefficients on SO_{it-1} capture the effect over a longer (two-year) period.

The main specification we take to the data controls for labour productivity, LP, capital intensity, KI, and log of wages, w, all of which also enter as lagged levels and yearly changes.¹⁴ Accordingly:

(3)
$$\Delta e_{it} = \alpha e_{it-1} + \mathbf{SO}_{it}\boldsymbol{\beta} + \mathbf{V}_{it}\boldsymbol{\gamma} + \eta_i + \varepsilon_{it},$$

 $^{^{14}{\}rm LP}$ is defined as log(value added/number of employees) while KI is log(physical capital/number of employees). Note that w refers to the log of wages.

where $\alpha \equiv \{\beta_1 - 1\}$, the vector $\mathbf{SO}_{it} \equiv \{SO_{it-1}^g, \Delta SO_{it}^g, SO_{it-1}^s, \Delta SO_{it}^s\}$ and the vector $\mathbf{V}_{it} \equiv \{\Delta LP_{it}, LP_{it-1}, \Delta KI_{it}, KI_{it-1}, \Delta w_{it}, w_{it-1}\}$. The specification above is estimated using firm-level fixed effects to control for productivity differences and any unobserved heterogeneity across firms.

Offshoring can affect labour in two ways. First, it can reduce the amount of employment to the extent that imported inputs replace a task previously done in-house. However, product improvements and efficiency gains due to the use of (new and better) imported inputs may have an expansion effect on firms' output and employment. In order to identify the complementarity or substitutability of offshoring with respect to labour, net of sales expansion due to rising productivity and increased demand, we will also estimate the specification above adding lagged log of output and growth of output in real terms as extra control variables:

(4)
$$\Delta e_{it} = \alpha e_{it-1} + \mathbf{SO}_{it}\boldsymbol{\beta} + \mathbf{V}_{it}\boldsymbol{\gamma} + \mathbf{Q}_{it}\boldsymbol{\delta} + \eta_i + \varepsilon_{it},$$

where $\mathbf{Q}_{it} \equiv \{q_{it-1}, \Delta q_{it}\}^{15}$ Hummels et al. (2014) use a similar approach to estimate the direct and aggregate (i.e., including productivity gains) effect of goods offshoring on wages in Denmark. But while Hummels et al. (2014) use levels of offshoring, we scale trade variables by firm size to eliminate the correlation in employment and imports that is unrelated to the offshoring decision.

B. Instruments

The key identification challenge we face in our empirical exercise is that productivity shocks and changes in the demand for the firms' products are likely to simultaneously affect employment and offshoring. For instance, there is undisputed evidence that highly productive firms are larger, export more products and import more inputs. Our specification tries to minimize these sources of endogeneity by including a rather rich set of control variables capturing the impact of

¹⁵Note that q is the log of real output Q.

technological changes, capital deepening and the cost of labour on employment. Nevertheless, estimating equation (3) with Ordinary Least Squares or FE may still result in biased parameter estimates if there are unobserved productivity changes and demand shocks that affect firms' offshoring intensity as well as their labour demand. Thus, we need to find firm-time varying instruments that are correlated with imports at the firm level but uncorrelated with labour demand.

To construct the instruments, we combine the firm-level trade structure in a base year with factors at the product-country level affecting the propensity to import (see, also, Hummels et al., 2014 for a similar approach). More precisely, our firm-time varying instrument Z_{it} for firm *i* in period *t* is constructed as $Z_{it} = \sum_{c,k} s_{ick} I_{ckt}$, where s_{ick} represents the share of imports of product *k* from country *c* by firm *i* in the base year–i.e., the pre-sample year or the first year that the firm enters the sample. The other component of the instrument I_{ckt} consists of factors affecting or reflecting changes in comparative advantage of country *c* in the production of product *k*. These include exchange rates, inflation, and world export supply in goods and services. These factors are discussed in more detail in subsequent sections.

EXCHANGE RATES. — The exchange rate between country c and Belgium in period t affects the costs of importing in euros (or in the Belgian franc for the period before 1999), thus affecting the offshoring intensity of Belgian firms from that country. However, the exchange rate itself is not related to the employment outcome of specific Belgian firms, making it a valid instrument. We obtained exchange rates from the International Financial Statistics database published by the IMF. Note that the exchange rates have only country-time variation, no product variation.

INFLATION RATES AND WAGES. — If inflation or wages in country c are higher than in Belgium, the imported products/services become more expensive, altering the optimal amount of imports. At the same time, these variables can be considered exogenous to the employment outcome of individual Belgian firms. As for the exchange rates, we obtained inflation rates from the International Financial Statistics Dataset, and they have country-time variation. Data on wages, were retrieved from the STructural ANalysis (STAN) database produced by the OECD and have a country-industry-time variation.

TRADE IN GOODS. — We follow Hummels et al. (2014) and construct so-called World Export Supply (WES), which is equal to total exports of product k by country c net of exports of product k by country c to Belgium. The idea is that WES captures changes in comparative advantage for the exporting country, arising from changes in the production price, product quality or variety. Subtracting the exports to Belgium from the total exports makes ensures that Belgian demand factors are filtered out of the instrument. For bilateral trade flows, we rely on the BACI dataset (Gaulier and Zignago, 2010), which is based on the COMTRADE dataset.¹⁶ From this database we can construct WES at the HS6 digit-yearcountry level.¹⁷ Other studies use variation in tariffs to construct a similar instrument. We experimented with tariffs from the UNCTAD Trains database, but the tariff instruments had little explanatory power in the first-stage regressions, probably due to the little variation in tariffs over the sample period.

TRADE IN SERVICES. — We construct a similar measure for World Export Supply, but for trade in services. Unfortunately, bilateral trade flows of services are not as detailed as those for trade in goods. The WTO reports trade in services, but only exports to the world for each country (and not bilateral trade flows). Therefore, we can not control for the exports to Belgium. However, given that

¹⁶The BACI dataset makes corrections to COMTRADE based on the observation that the reported imports of country j from country c should be the same as the reported exports of country c to country j, after taking into account that import values are reported CIF (cost, insurance and freight) and export values are reported FOB (free on board).

 $^{^{17}}$ To control for HS6 codes changing over time, we follow the concordance procedure explained in Van Beveren et al. (2012).

Belgium is a small country, it is unlikely to affect the aggregate export patterns of the partner countries. Note that the product level is very coarse however, as we can only separately observe exports of commercial services and transport services throughout the sample period.

The different variables described above have country-time or country-timeindustry variation. As mentioned above, we obtain firm-year level instruments by taking a weighted sum of these factors with the share of the import value of product k from country c by firm i in the base year as weight. The idea is that a firm i that sources a particular input k from country c is likely to keep on buying this input k from country c over a long period of time because, for example, the product is a particularly good fit for the firm or because there are fixed costs associated with switching countries to buy the inputs (Hummels et al., 2014). Over time, the exporting country's comparative advantage in a product can change, and these changes will affect the supply decisions of firms already importing from this country more than other firms.

In the empirical specification, we use the exchange rate, trade in goods and wages as instruments for offshoring of goods, and the exchange rate, trade in services and inflation for offshoring of services. Econometric tests reported in the following section confirm that our instruments are highly correlated with the import share while being orthogonal to the changes in firm level employment.

IV. Results

Before investigating the impact of offshoring on employment, we start exploring the relationship among total employment, national intermediary inputs, and offshoring. More precisely, we regress the growth rate of employment on the growth rate of the other inputs-namely, domestically purchased¹⁸ and imported inputs:

¹⁸Domestically purchased inputs are obtained by subtracting firm level imports, excluding imports of capital goods according to the BEC classification, from total reported spending on intermediate goods and services.

$$\Delta e_{it} = \beta_m \Delta m_{it} + \beta_o \Delta o_{it} + u_{it},$$

where Δe_{it} is the growth rate of average yearly employment of firm *i* between period t-1 and t, and Δm_{it} and Δo_{it} are, respectively, the growth rate of domestically purchased inputs and total offshoring (i.e. goods and services together). Similarly, we estimate a model with Δm_{it} on the left-hand side (LHS) and Δe_{it} on the right-hand side (RHS). Each of these two specifications is estimated a second time by adding Δq_{it} as a control variable. This exercise, exploratory in nature, is aimed at understanding the strength of complementarity and substitutability between offshoring, on one side, and labour and national intermediary inputs, on the other side.

Two interesting facts emerge from the estimates reported in Table 3. When we do not control for real output growth Δq , we find a positive relationship between the growth of offshoring and growth of employment, while we find a negative relationship between the growth of national inputs and offshoring. Once we include Δq on the RHS, we uncover a negative relationship between Δe and Δo in the services sector but not in the manufacturing sector. The point estimate for the services sector is, however, very small in absolute value. Overall, these findings suggest that most of the offshoring involves looking abroad for new suppliers of goods and services previously bought on national markets, rather than the (much feared) relocation of parts of the production process to low-income countries. This is also consistent with the fact that in Belgium (as in other European countries), most of the trading of goods and services is within the EU.

A. Main Results

We now turn to the main results obtained from estimating equations (3) and (4). As explained in the empirical section, all specifications include firm fixed effects and a number of control variables, such as (growth rates and lagged levels of) labour productivity, capital intensity and the wage rate. ¹⁹ The lagged level of dependent variable is also included on the RHS to control for potential mean reversion in firm employment growth.²⁰ Hereafter, we first discuss the results for total employment and then those for different levels of education.

TOTAL EMPLOYMENT. — Table 4 reports the estimated coefficients for the manufacturing sector on the left panel and for the services sector on the right panel. In each case, we first show the firm fixed effect (FE) results and then the FE-IV results, where we instrument for the four offshoring variables.

The diagnostics for the IV specifications show that the Kleibergen-Paap test statistic of weak instruments (Kleibergen and Paap, 2006) strongly rejects the null of underidentification in all specifications.²¹ At the same time, the p-values of the Hansen test of overidentifying restrictions support the idea that instruments are exogenous for the service sector.²² For the manufacturing industries, there appears to be a problem with the exogeneity of the instruments for the specification that include the control for output \mathbf{Q}_{it} . We note that in the check of robustness reported at the end of this section, we obtain a Hansen p-value of 0.14 when using data for the period 1998-2005 (cf column (6) in Table 9).²³.

In the manufacturing sector, when we do not control for output growth nor we instrument for the offshoring variables, both the change in goods offshoring and the lagged level of goods offshoring are positively related to employment growth

 $^{^{19}}$ The coefficients of most of the control variables (not reported in the tables but available upon request) take the expected signs and are very precisely estimated. For instance, the coefficients on growth and the lagged level of the hourly wage are both negative and significant at the 1 per cent level. 20 See, for example, Sutton (1997) for a survey on the relation between firm size and employment growth.

²¹Similarly, the Angrist and Pischke (2009) F-test of weak and underidentification (not reported but available upon request), strongly rejects underidentification of each single endogenous variable, with p-values below 0.001 in all the specifications presented in this Section.

²²The endogeneity test is calculated as the difference between the Hansen statistic for the regression where the offshoring variables are treated as endogenous and the Hansen statistic where they are assumed to be exogenous. This difference has a χ^2 distribution with degrees of freedom equal to the number of variables tested for exogeneity. Under homoskedasticity, this test is numerically equal to a Hausman test statistic. Baum et al. (2007)

 $^{^{23}}$ More generally, we note that, by changing the combination of instruments, we can obtain higher p-values of the Hansen test with no major changes in the point estimates of our offshoring variables. We believe that it is cleaner to use the same set of instruments across all specifications, even if in few instances, we get low Hansen p-value

while we do not find any significant effect of service offshoring on employment growth. The coefficients on SO_{it-1}^g and on ΔSO_{it-1}^g are not statistically different. As mentioned before, the estimated effects of offshoring are a combination of a scale effect and a technology effect. The scale effect could stem from imports of intermediate products increasing firm-level productivity because of learning, variety or quality effects Amiti and Konings (2007), thus boosting competitiveness and increasing sales and, in turn, employment growth.²⁴ The fact that the coefficient for ΔSO_{it}^g is positive in column (1) but negative in column (2) suggests that in the short-run, there is a certain degree of substitutability between goods offshoring and labour, but this effect is more than offset by an expansion in output.

Even after including the different controls and firm FE in the regression, there could still be an endogeneity problem in the form of firms' specific demand or productivity shocks that increase simultaneously labour demand and the share of imported intermediary inputs. We use IV regression to control for this endogeneity problem. The results in column (3) show that our IV strategy leads to a drammatic increase in the point estimates for all offshoring variables: we now find a large positive effect of both goods and services offshoring on employment.

The short-run coefficient on ΔSO_{it-1}^g is significantly larger than the coefficient on SO_{it-1}^g , a common feature across most of the specifications that will be presented in the rest of this Section. This result suggests that the positive effects of offshoring are often halved when considering a longer period, possibly because of the longer time required to adjust the labour force in Belgium. This interesting finding confirms the importance of using an empirical framework that allows us to assess the effect of offshoring at different points in time. Interestingly enough, the two "long-term" coefficients on the lagged offshoring variables are no longer significant when we control for output growth. This is consistent with the idea

 $^{^{24}\}mathrm{Note}$ that the increase in sales must be higher than the increase in imports to have an increase in the share of offshoring.

that the offshoring does not lead to a substitution of tasks previously performed in-house; rather, it replaces domestic providers with suppliers located abroad.

The results for the services sector on the right panel show a more complex picture. Similar to the manufacturing sector, an increase in the offshoring of goods is related to an increase in employment, and this effect disappears when we include controls for output growth. This, again, can be explained by service firms replacing national intermediate goods with foreign goods that have a better price/quality ratio. Offshoring of services is, instead, negatively related to employment growth in the fixed effects specifications. When instrumenting for the offshoring variables, we estimate a positive effect of ΔSO^s on employment growth, but this is a short-lived effect that disappears after one period, given that the coefficient on lag offshoring is insignificant. The change in the coefficient for ΔSO^s is consistent with the idea that firms face labour-saving technical changes that, at the same time, facilitate offshoring of services. Finally, results in the last column show that once we control for changes in output, service offshoring makes firms in the service sector less labour intensive. However, it is important to note that the size of the effect remains relatively small. For example, a firm that would increase its share of service offshoring from 0 to the average 0.033, would see a reduction in employment of around 1%, holding output fixed. This limited effect of offshoring is mainly driven by the relatively low level of service offshoring during our sample period (1996-2005). Over the last decade, services trade has continued to grow at a fast pace. If the marginal effect of service offshoring on employment growth has remained stable after our sample period, this implies that the contribution of service offshoring to employment losses can be substantially higher. However, the relation between service offshoring and employment could as well have changed over the last decade, making it difficult to extrapolate our findings to more recent years.

EDUCATION LEVEL. — The fear of offshoring is not only driven by total job losses, but also by concerns that offshoring favours one group of employees over the other, based on their education/skill level. In this subsection, we look at the impact of offshoring on low- and high-educated employees.

Tables 5 and 6 report the IV results for manufacturing and services, respectively. The first two columns repeat the specification used in the previous section for the sake of comparison.

Columns (3) and (5) in Table 5 shows that the positive effects of offshoring on employment growth in the manufacturing sector are shared by low- and highskilled workers. Given our definition of the dependent variable as the change in employment of a particular skill type divided by total employment, the coefficient on the offshoring variable also varies with the importance of the labour type in total employment. So, given that the fraction of low-skilled workers in manufacturing industries is approximately four times as large as the fraction of high-skilled workers, the coefficient on offshoring for the low-skilled should also be four times as large as that for the high-skilled in order to keep the relative shares of the different types at the same level. However, the coefficients for the high-skilled are relatively larger when not controlling for output, indicating that the firms are becoming more skill-intensive. This may explain why the only coefficient that it is marginally significant when controlling for output is for the skilled workers in column (6).

The results for the services sector are included in Table 6. The positive effect of goods offshoring on employment growth when we do not condition on output growth is shared by both low-educated and high-educated workers. For service offshoring, the negative effect on employment is entirely due to high-skilled employment. The results indicate that the offshored activities in Belgium are executed mainly by highly-educated employees. This is one of our key findings, as it offers the first large-scale evidence using firm-level data that service offshoring may threaten the job security of higher-educated workers. Given that the vast majority of imported services originate in other developed economies, the negative impact is not due to trading with low-wage countries. We will provide further evidence on this issue in the next subsection.

B. Robustness Checks and Extensions

In this part, we check the robustness of the results when using a "narrow" definition of offshoring and when reducing the sample period to 1998-2005.²⁵. Moreover we also explore whether the results above are driven by imports from OECD or non-OECD countries.²⁶

TRANSPORT SERVICES EXCLUDED/ NARROW DEFINITION OF GOODS OFFSHORING. — In the literature, there is a discussion of whether to define offshoring narrowly or more broadly. ²⁷ This rests on the idea that the closer the imported inputs are to the outputs produced by the firm, the more likely it is that within-firm labour could have produced those inputs. Moreover, up till now, we have included transport services in out measure for service offshoring. However, especially for the manufacturing sector, these are likely to relate strongly to exports of goods. In Table 7, we show results for the manufacturing sector using a "narrow" definition of goods offshoring and not including transport in service offshoring. Results are very similar to those presented in the previous section; in particular, offshoring of goods is confirmed to have a large positive impact on employment of all types of workers, but for workers with low education, this effect disappears when we control for output.

 $^{^{25}}$ As mentioned in the Data section, the threshold at which firms are asked to report intra-EU imports and exports changed in 1998, so focusing on the 1998-2005 period can soften problems due to changes in data reporting conditions.

 $^{^{26}}$ We have also experimented with alternative specifications, and obtained similar results, when: (i) including a control for exports; (ii) adding an aggregate measure of import competition at industry-level; and (iii) using only lagged offshoring instead of our richer dynamic framework.

 $^{^{27}}$ Note that if we were to apply the definition of "narrow" offshoring to services, the value of service offshoring would be equal to zero for manufacturing firms.

CORE SERVICES. — We apply similar ideas to the services sector and we repeat the same analysis, but we now include only the core services—that is, the offshoring of services that correspond to the main activity of the firm. Results reported in Table 8 confirm the findings in Table 6, most notably the fact that it is only for workers with a high level of education that (i) there is a negative overall effect of offshoring (see the coefficient on SO_{it-1}^s in column (5)); and (ii) there is a strong substitution effect between labour and services offshoring when we control for changes in output (see column(6)).

PERIOD 1998-2005. — We re-estimate equations (3) and (4) using data for the period 1998-2005 as in the first years of our data set the thresholds to report trade data have changed. Moreover, the reporting of employment changed in 1996. Table 9 shows that results are very similar to those reported in Tables 5 and 6 above. The diagnostic tests for IV regressions confirm again that excluded instruments are highly correlated with the endogenous offshoring variables; at the same time, we now obtain higher p-values of the Hansen statistics, thus giving further support to the validity of our IV strategy.

INCLUDING ORIGIN OF IMPORTS. — Several empirical works have found that the negative impact of goods offshoring on manufacturing sector employment is driven by imports from developing countries—China in particular. For instance, Pierce and Schott (2016) find: "Industries where the threat of tariff hikes declines the most experience more severe employment losses along with larger increases in the value of imports from China and the number of firms engaged in China-U.S. trade." In the present context, it is interesting to assess whether our two main results (i.e., the positive effect of goods offshoring on employment for all types of workers and the negative impact of services offshoring on highly-educated workers) change according to the origin of the imported inputs. For this, we divide our measure of offshoring intensity between offshoring from OECD countries and from

non-OECD countries. Three interesting facts emerge from the results reported in Table 10. First, imported goods from both OECD and non-OECD countries have a positive impact on workers with low education, especially in the manufacturing sectors, when not controlling for output growth. This finding suggests that for the workforce in a particular firm, goods offshoring from developing countries does not represent a threat. However, when controlling for output growth, offshoring to non-OECD countries appears to have a long-term negative impact on employment growth, indicating that the production process becomes less labor intensive. When distinguishing between low and high skilled workers, the coefficient remains negative but becomes insignificant for both categories. Second, the positive impact of goods offshoring on employees with high education is driven by imports from OECD countries. This result is probably driven by the fact that the imports from Western economies represent a better match with the skills of this group of workers. Finally, the negative effect of services offshoring on higher-educated workers is driven by trading with OECD countries.

V. Conclusion

The impact of globalisation on domestic labour markets has been the subject of a large debate within the academic arena and among the general public. While the focus has historically been on goods, offshoring of services has recently attracted more and more interest because advances in information and communication technology have led to a remarkable increase in the tradability of services.

Using a newly constructed dataset that contains firm-level information on imports of both goods and services, this study produces fresh evidence on the impact of offshoring on the employment outcome in manufacturing industries and services sectors. We deal with the standard endogeneity problem of offshoring by constructing firm-level instrumental variables based on high-quality information on exchange rates, inflation, wages and international trade flows. Diagnostic tests presented in the empirical section show that our IVs are both highly correlated with offshoring and exogenous with respect to the possible demand and productivity shocks buried in the residual.

We find that employment growth of both workers with low and high education is positively related to goods offshoring in the manufacturing sector. Our empirical specification allows us to uncover a substantial reduction in this positive impact over time, with a coefficient on lagged offshoring often halved compared to the (short-run) effect in period t. Moreover, we find that this positive relationship disappears when controlling for scale effects. For the services sectors, our key finding is that there is a negative effect of service offshoring on the employment outcome of workers with higher education. To the best of our knowledge, this is the first large-scale study with detailed information on services offshoring that showing that globalisation may threaten the job security of higher-educated workers. We show that this result is driven by trading within OECD countries, which is not unsurprising given that the vast majority of trade in services is with these countries. This result contrasts the findings of previous studies, in which trading from low-wage countries–China in particular, was the main culprit for the loss of blue-collar jobs in the manufacturing industries.

The fact that service offshoring represents a small percentage of firms' turnover implies that its impact on the employment of higher-educated workers is not large in the period we study. However, the nature of our findings are very important if we take into consideration that service offshoring has grown at a faster pace than goods offshoring in the last decade, a trend that may even increase given that ongoing trade negotiations are increasingly focused on facilitating market access in areas such as licensing, telecommunications, e-commerce and financial services.

A limitation of our work is that we only look at the effects of offshoring on individual firms. To assess the aggregate impact of service offshoring, one would need to take into account the impact on domestic suppliers and competitors, which could experience employment losses and could be more likely to exit the market. Determining these effects lies outside the scope of the current paper and we leave this for future research.

REFERENCES

- Acemoglu, Daron, David H. Autor, David Dorn, Gordon Hanson, and Brendan Price, "Import Competition and the Great US Employment Sag of the 2000s," NBER Working Paper, 2014, 20395.
- Amiti, Mary and Jozef Konings, "Trade Liberalization, Intermediate Inputs, and Productivity: Evidence from Indonesia," 2007, 97 (5), 1611–1638. The American Economic Review.
- _____ and Shang-Jin Wei, "Fear of service outsourcing: Is it justified?," *Economic Policy*, 2005, *20* (42), 308–347. Fear of service outsourcing: Is it justified?
- **and** _____, "Service offshoring, productivity and employment: evidence from the United States," *CEPR Discussion Paper*, 2006, 5475.
- Andersson, Linda, Patrik Karpaty, and Selen Savsin, "Firm-level effects of offshoring of materials and services on relative labor demand," *Review of World Economics*, 2016, 152 (2), 321–350.
- Angrist, J. D. and Jorn-Steffen Pischke, Mostly harmless econometrics : an empiricist's companion, Princeton: Princeton University Press, 2009.
- Autor, David H., David Dorn, and Gordon H. Hanson, "The China Syndrome: Local Labor Market Effects of Import Competition in the United States," American Economic Review, 2013, 103 (6), 2121–68.
- Baum, Christopher F., Mark E. Schaffer, and Steven Stillman, "Enhanced routines for instrumental variables/generalized method of moments estimation and testing," *The Stata Journal*, 2007, 7 (4), 465–506.
- Bernard, Andrew B., J. Bradford Jensen, and Peter K. Schott, "Survival of the best fit: Exposure to low-wage countries and the (uneven) growth of U.S. manufacturing plants," *Journal of International Economics*, 2006, 68 (1), 219–237.
- Blinder, Alan S., "Offshoring: The Next Industrial Revolution?," in "Foreign Affairs," Vol. 85, Foreign Affairs, 2006, pp. 113–128.

- Breinlich, Holger and Chiara Criscuolo, "International trade in services: A portrait of importers and exporters," *Journal of International Economics*, 2011, 84.
- Crino, Rosario, "Offshoring, Multinationals and Labour Market: A Review of the Empirical Literature," *Journal of Economic Surveys*, 2009, 23 (2), 197–249.
- _____, "Employment effects of service offshoring: Evidence from matched firms," *Economics Letters*, 2010, 107 (2), 253–256.
- _____, "Service Offshoring and White-Collar Employment," The Review of Economic Studies, 2010, 77 (2), 595–632.
- , "Service Offshoring and the Skill Composition of Labour Demand," Oxford Bulletin of Economics and Statistics, 2012, 74 (1), 20–57.
- Feenstra, Robert C. and Gordon H. Hanson, "The Impact of Outsourcing and High-Technology Capital on Wages: Estimates For the United States, 19791990," *The Quarterly Journal of Economics*, 1999, 114 (3), 907–940.
- _____ and _____, "Global Production Sharing and Rising Inequality: a Survey of Trade and Wages," in K Choi and James Harrigan, eds., *Handbook of International Economics*, Oxford: Blackwell Publishing, 2003.
- Francois, Joseph and Bernard Hoekman, "Services Trade and Policy," Journal of Economic Literature, 2010, 48 (3), 642–92.
- Gaulier, Guillaume and Soledad Zignago, "BACI: International Trade Database at the Product-Level. The 1994-2007 Version," CEPII Working Paper, 2010, 2010-23.
- Geishecker, Ingo and Holger Görg, "Services offshoring and wages: evidence from micro data," Oxford Economic Papers, 2013, 65 (1), 124–146.
- Goldberg, Pinelopi Koujianou, Amit Kumar Khandelwal, Nina Pavcnik, and Petia Topalova, "Imported Intermediate Inputs and Domestic Product Growth: Evidence from India," *The Quarterly Journal of Economics*, 2010, 125 (4), 1727– 1767.

- Görg, Holger and Aoife Hanley, "Labour demand effects of international outsourcing: Evidence from plant-level data," International Review of Economics & Finance, 2005, 14 (3), 365–376.
- Grossman, G.M. and E. Rossi-Hansberg, "Trading tasks: A simple theory of offshoring," *American Economic Review*, 2008.
- Halpern, L., M. Koren, and A. Szeidl, "Imported inputs and productivity," Unpublished manuscript, 2011.
- Hijzen, Alexander, Mauro Pisu, Richard Upward, and Peter W. Wright, "Employment, job turnover, and trade in producer services: UK firm-level evidence.," Canadian Journal of Economics/Revue canadienne d'conomique, 2011, 44 (3), 1020– 1043.
- Hummels, David, Rasmus Jrgensen, Jakob Munch, and Chong Xiang, "The Wage Effects of Offshoring: Evidence from Danish Matched Worker-Firm Data," American Economic Review, 2014, 104 (6), 1597–1629.
- Kleibergen, Frank and Richard Paap, "Generalized reduced rank tests using the singular value decomposition," *Journal of Econometrics*, 2006, 133 (1), 97–126.
- Krugman, Paul, "Trade and Wages, Reconsidered," Brookings Papers on Economic Activity, 2008, 39, 103–154.
- Mion, Giordano and Linke Zhu, "Import competition from and offshoring to China: A curse or blessing for firms?," *Journal of International Economics*, 2013, 89, 202–2015.
- Monarch, Ryan, Jooyoun Park, and Jagadeesh Sivadasan, "Domestic gains from offshoring? Evidence from TAA-linked U.S. microdata," *Journal of International Economics*, 2017, 105, 150 – 173.
- Pierce, Justin R. and Peter K. Schott, "The Surprisingly Swift Decline of US Manufacturing Employment," American Economic Review, 2016, 106 (7), 1632–62.
- Sutton, J., "Gibrat's legacy," Journal of Economic Literature, 1997, 35 (1), 40–59.

- Van Beveren, Ilke, Andrew B. Bernard, and Hylke Vandenbussche, "Concording EU trade and production data over time," *Tuck School of Business, mimeo*, 2012.
- Wright, Greg C., "Revisiting the employment impact of offshoring," European Economic Review, 2014, 66, 63–83.





Full Sample, Transport Services Included

FIGURE 1. EVOLUTION TRADE IN GOODS AND SERVICES





Offshoring Intensity per Sector

Average share of imports in turnover; Goods excludes capital goods

VII. Tables

		1996			2005	
Type of Service	Value	Share	$\%~{\rm Firms}$	Value	Share	%Firms
Transport Services	2,054	61.5%	13.6%	3,771	45.9%	14.8%
Business Services	788	23.6%	13.8%	2,411	29.4%	21.5%
IT Services	189	5.7%	5.6%	965	11.8%	8.2%
Communication Services	228	6.8%	2.0%	910	11.1%	2.8%
Financial Services	80	2.4%	11.2%	151	1.8%	12.2%
Total Trade in Services	3,339	100%	28.2%	8,208	100%	38.8%
Total Trade in Goods	29,766		57.2%	56,219		51.6%

TABLE 1—IMPORTS OF DIFFERENT SERVICE TYPES (VALUES IN MILLION EUROS)

TABLE 2—Summary Statistics

Variable	Mean	p50	sd	Obs
Manufacturing				
Turnover $(\times 1000 \in)$	44,932	$11,\!377$	181,515	$24,\!489$
Employment	153	56	424	$24,\!489$
Offshoring of Goods ($\times 1000 \in$)	$13,\!309$	$1,\!654$	88,390	24,489
Offshoring of Services ($\times 1000 \in$)	552	0	7,163	24,489
Share Offshoring Goods	0.194	0.145	0.189	24,489
Share Offshoring Services	0.005	0	0.021	$24,\!489$
Employment Growth	0.45%	0.0%	16.4%	$24,\!489$
Services				
Turnover (×1000 \in)	21,744	$5,\!184$	$117,\!440$	20,161
Employment	156	27.1	1,389	20,161
Offshoring of Goods (×1000€)	450	0	6,385	20,161
Offshoring of Services ($\times 1000 \in$)	1,702	0	14,977	20,161
Share Offshoring Goods	0.013	0	0.067	20,161
Share Offshoring Services	0.033	0	0.077	20,161
Employment Growth	3.6%	1.7%	23.1%	20,161

		Manuf	acturing			Sei	rvice	
	(1)	(2)	(3)	(4)	(1)	(2)	(3)	(4)
	Empl	Empl	Nat Mater.	Nat Mater.	Empl	Empl	Nat Mater.	Nat Mater.
Growth National Materials	0.121^{***}	-0.0181***			0.191^{***}	-0.0406***		
	(0.00496)	(0.00591)			(0.00783)	(0.00954)		
Growth Tot Employment			0.432^{***}	-0.0398***			0.368^{***}	-0.0554***
			(0.0171)	(0.0128)			(0.0149)	(0.0129)
Growth Offshoring	0.0670^{***}	-0.00095	-0.0505***	-0.240***	0.0363^{***}	-0.00831^{*}	-0.0654^{***}	-0.115^{***}
	(0.00315)	(0.00364)	(0.00621)	(0.00531)	(0.00483)	(0.00468)	(0.00749)	(0.00571)
Growth Output		0.318^{***}		1.077^{***}		0.436^{***}		0.842^{***}
		(0.0115)		(0.0126)		(0.0135)		(0.0139)
R-squared	0.233	0.392	0.020	0.073	0.0709	0.209	0.083	0.441
Ν	21431	21409	21852	21784	14161	14121	14161	14121
^a Robust standard errors in p	arentheses. *1	p < 0.10, * * p	< 0.05, * * *p	< 0.01				

TABLE 3—SUBSTITUTABILITY BETWEEN INPUTS

^b Regression of respectively growth in employment and growth in domestically purchased intermediates on growth in other inputs.

		Manufac	turing			Servi	ces	
	(1)	(2)	(3)	(4)	(5)	(9)	(2)	(8)
	FE	FE	IV-FE	IV-FE	FE	FE	IV-FE	IV-FE
$ShOf_{it-1}^{g}$	0.111^{***}	-0.033	0.737^{***}	0.166	0.134^{*}	-0.038	1.451^{***}	0.285
	(0.02)	(0.02)	(0.12)	(0.12)	(0.07)	(0.07)	(0.26)	(0.20)
$\Delta ShOf_{it}^g$	0.084^{***}	-0.069***	1.383^{***}	0.347^{*}	0.188^{***}	-0.032	1.638^{***}	0.117
	(0.02)	(0.02)	(0.20)	(0.19)	(0.07)	(0.06)	(0.43)	(0.35)
$ShOf_{it-1}^s$	0.070	0.050	0.616	0.059	-0.100^{**}	-0.098***	0.033	-0.281**
	(0.12)	(0.11)	(0.38)	(0.28)	(0.05)	(0.04)	(0.15)	(0.12)
$\Delta ShOf_{it}^s$	0.025	0.020	0.839^{**}	-0.078	-0.120^{***}	-0.144***	0.435^{***}	-0.286^{*}
	(0.09)	(0.07)	(0.42)	(0.31)	(0.04)	(0.04)	(0.19)	(0.15)
Controls for Output		Х		Х		X		Х
Obs	24489	24489	24489	24489	20161	20161	20161	20161
Test $ShOf_{it-1}^g = \Delta ShOf_{it}^g$.184	.060	000.	.220	.329	006.	.608	.575
Test $ShOf_{it-1}^s = \Delta ShOf_{it}^s$.634	.746	.558	.630	.633	.207	.016	.972
Underidentification			000.	000.			.000	000.
Endog. Test			000.	.245			.000	.257
Hansen p -val			.525	.053			.411	.664
$\frac{a}{b}$ Robust standard errors in parer	theses. $*p < \frac{1}{2}$	0.10, * * p < 0	0.05, * * * p < 0.05	0.01	Ē	4:		4

TABLE 4—EMPLOYMENT GROWTH AND OFFSHORING

An spectructions include controls for wages, labor productivity and capital intensity. The controls enter in the same way as the offshoring variables, i.e. in the specification where offshoring enters in lagged levels as well as in growth, the control variables also enter in both lagged levels and growth. The same holds for controlling for output. If the specification controls for output, output enters in the same way as offshoring. Finally, the lagged level of employment is included as well. All specifications include firm fixed effects.

	Tot	al	Low S	killed	High	Skilled
	(1)	(2)	(3)	(4)	(5)	(6)
	IV-FE	IV-FE	IV-FE	IV-FE	IV-FE	IV-FE
$ShOf_{it-1}^{g}$	0.737***	0.166	0.474***	0.040	0.205***	0.100*
	(0.12)	(0.12)	(0.09)	(0.10)	(0.05)	(0.06)
$\Delta ShOf_{it}^g$	1.383***	0.347^{*}	0.900***	0.141	0.331***	0.116
	(0.21)	(0.19)	(0.16)	(0.16)	(0.09)	(0.10)
$ShOf_{it-1}^s$	0.616	0.059	0.285	-0.123	0.233	0.118
	(0.38)	(0.28)	(0.28)	(0.23)	(0.15)	(0.13)
$\Delta ShOf_{it}^s$	0.839**	-0.078	0.157	-0.489*	0.436**	0.230
	(0.42)	(0.31)	(0.32)	(0.26)	(0.17)	(0.16)
Controls for Output		Х		Х		Х
Obs	24489	24489	24489	24489	24489	24489
$ShOf_{it-1}^g = \Delta ShOf_{it}^g$.000	.220	.001	.418	.125	.836
$ShOf^s_{it-1} = \Delta ShOf^s_{it}$.558	.630	.6631428	.130	.186	.424
Underidentification	.000	.000	.000	.000	0	$0.00E{+}00$
Endog. Test	.000	0.245	.000	.341	0	0.050
Hansen $p-val$.539	.053	.411	.113	.157	0.041

TABLE 5—EMPLOYMENT GROWTH AND OFFSHORING IN MANUFACTURING SECTORS; LOW VERSUS HIGH EDUCATED

 $\overline{^a}$ Robust standard errors in parentheses. $*p < 0.10, **p < 0.05, ** \overline{*p < 0.01}$

^b All specifications include controls for wages, labor productivity and capital intensity. The controls enter in the same way as the offshoring variables, i.e. in the specification where offshoring enters in lagged levels as well as in growth, the control variables also enter in both lagged levels and growth. The same holds for controlling for output. If the specification controls for output, output enters in the same way as offshoring. Finally, the lagged level of employment is included as well. ^c All specifications include firm fixed effects.

	То	tal	Low Sl	killed	High	Skilled
	(1)	(2)	(3)	(4)	(5)	(6)
	IV-FE	IV-FE	IV-FE	IV-FE	IV-FE	IV-FE
$ShOf_{it-1}^{g}$	1.451***	0.285	0.590***	-0.012	0.635***	0.174
	(0.26)	(0.20)	(0.17)	(0.16)	(0.15)	(0.14)
$\Delta ShOf_{it}^g$	1.638^{***}	0.117	0.693**	-0.118	0.773***	0.176
	(0.43)	(0.35)	(0.30)	(0.28)	(0.26)	(0.24)
$ShOf_{it-1}^s$	0.033	-0.281**	0.237**	0.075	-0.194**	-0.321***
	(0.15)	(0.12)	(0.10)	(0.09)	(0.10)	(0.09)
$\Delta ShOf_{it}^s$	0.435^{**}	-0.286*	0.255*	-0.110	0.072	-0.210**
	(0.19)	(0.15)	(0.13)	(0.12)	(0.12)	(0.10)
Controls for Output		Х		Х		Х
Obs	20161	20161	20161	20161	20161	20161
$ShOf_{it-1}^g = \Delta ShOf_{it}^g$.608	.575	.690	.655	.530	.992
$ShOf^s_{it-1} = \Delta ShOf^s_{it}$.016	.972	.881	.085	.010	.235
Underidentification	.000	.000	.000	.000	.000	.000
Endog. Test	.000	.257	.001	.695	.000	.036
Hansen $p-val$.411	.663	.726	.569	.560	.904

TABLE 6—Employment growth and offshoring in services sectors; Low versus high educated

^{*a*} Robust standard errors in parentheses. *p < 0.10, **p < 0.05, ***p < 0.01

 b All specifications include controls for wages, labor productivity and capital intensity. The controls enter in the same way as the offshoring variables, i.e. in the

specification where offshoring enters in lagged levels as well as in growth, the control variables also enter in both lagged levels and growth. The same holds for controlling for output. If the specification controls for output, output enters in the same way as offshoring. Finally, the lagged level of employment is included as well.

^c All specifications include firm fixed effects.

	Tot	al	Low Sl	killed	High S	killed
	(1)	(2)	(3)	(4)	(5)	(6)
	IV-FE	IV-FE	IV-FE	IV-FE	IV-FE	IV-FE
$ShOf_{it-1}^{g}$	1.355***	0.382	0.864***	0.144	0.389***	0.214*
	(0.26)	(0.24)	(0.19)	(0.19)	(0.10)	(0.12)
$\Delta ShOf_{it}^{g}$	4.076***	1.195^{**}	2.600***	0.523	0.995***	0.413
	((0.68)	(0.57)	(0.51)	(0.48)	(0.29)	(0.29)
$ShOf_{it-1}^s$	1.273	0.159	0.547	-0.238	0.530*	0.290
	(0.78)	(0.48)	(0.55)	(0.38)	(0.28)	(0.23)
$\Delta ShOf^s_{it}$	2.282***	0.207	0.814	-0.661	1.000***	0.530^{**}
	(0.85)	(0.53)	(0.61)	(0.42)	(0.31)	(0.26)
Controls for Output		Х		Х		Х
Obs	24518	24518	24518	24518	24518	24518
$ShOf_{it-1}^g = \Delta ShOf_{it}^g$.000	.093	.000	.348	.022	.436
$ShOf^s_{it-1} = \Delta ShOf^s_{it}$.183	.921	.627	.288	.085	.303
Underidentification	.000	.000	.000	.000	.000	.000
Endog. Test	.000	.139	.000	.225	.000	.040
Hansen $p-val$.350	.109	.232	.134	.288	.076

TABLE 7—Employment growth and offshoring in manufacturing sectors; Narrow goods offshoring and service offshoring without transport services

^{*a*} Robust standard errors in parentheses. *p < 0.10, **p < 0.05, ***p < 0.01^{*b*} All specifications include controls for wages, labor productivity and capital intensity. The controls enter in the same way as the offshoring variables, i.e. in

intensity. The controls enter in the same way as the offshoring variables, i.e. in the specification where offshoring enters in lagged levels as well as in growth, the control variables also enter in both lagged levels and growth. The same holds for controlling for output. If the specification controls for output, output enters in the same way as offshoring. Finally, the lagged level of employment is included as well.

 $^{c}\,\mathrm{All}$ specifications include firm fixed effects.

	То	tal	Low Sl	killed	High	Skilled
	(1)	(2)	(3)	(4)	(5)	(6)
	IV-FE	IV-FE	IV-FE	IV-FE	IV-FE	IV-FE
$ShOf_{it-1}^{g}$	1.405***	0.269	0.561***	-0.025	0.626***	0.175
	(0.25)	(0.20)	(0.17)	(0.15)	(0.15)	(0.14)
$\Delta ShOf_{it}^{g}$	1.613^{***}	0.100	0.642**	-0.165	0.794***	0.198
	(0.42)	(0.35)	(0.30)	(0.28)	(0.26)	(0.24)
$ShOf_{it-1}^s$ Core	0.048	-0.537**	0.431**	0.130	-0.359*	-0.598***
	(0.30)	(0.23)	(0.20)	(0.18)	(0.18)	(0.17)
$\Delta ShOf_{it}^s$ Core	1.190^{***}	-0.613*	0.634**	-0.284	0.277	-0.429*
	(0.46)	(0.36)	(0.32)	(0.29)	(0.28)	(0.26)
Controls for Output		Х		Х		Х
Obs	20275	20275	20275	20275	20275	20275
$ShOf_{it-1}^g = \Delta ShOf_{it}^g$.556	.560	.749	.545	.433	.908
$ShOf^s_{it-1} = \Delta ShOf^s_{it}$.005	.815	.478	.116	.011	.457
Underidentification	.000	.000	.000	.000	.000	.000
Endog. Test	.000	.185	.003	.703	.000	.025
Hansen p -val	.481	.655	.728	.574	.631	.899

TABLE 8—Employment growth and offshoring in services sectors; offshoring of core services

^{*a*} Robust standard errors in parentheses. *p < 0.10, **p < 0.05, **p < 0.01

^b All specifications include controls for wages, labor productivity and capital intensity. The controls enter in the same way as the offshoring variables, i.e. in the

specification where offshoring enters in lagged levels as well as in growth, the control variables also enter in both lagged levels and growth. The same holds for controlling for output. If the specification controls for output, output enters in the same way as offshoring. Finally, the lagged level of employment is included as well.

^c All specifications include firm fixed effects.

			Manufae	cturing					Serv	ices		
	(1)	(2)	(3)	(4)	(5)	(9)	(2)	(8)	(6)	(10)	(11)	(12)
	Tot	al	Low S.	killed	High S.	killed	Tot	tal	Low Sk	tilled	High 5	Skilled
	IV-FE	IV-FE	IV-FE	IV-FE	IV-FE	IV-FE	IV-FE	IV-FE	IV-FE	IV-FE	IV-FE	IV-FE
$ShOf_{it-1}^{g}$	0.711^{***}	0.140	0.458^{***}	0.028	0.207^{***}	0.094	1.441^{***}	0.231	0.542^{***}	-0.071	0.650^{***}	0.166
	(0.13)	(0.13)	(0.10)	(0.10)	(0.05)	(0.06)	(0.29)	(0.23)	(0.20)	(0.18)	(0.17)	(0.16)
$\Delta ShOf_{it}^{g}$	1.292^{***}	0.289	0.884^{***}	0.154	0.276^{***}	0.059	1.510^{***}	0.189	0.572^{*}	-0.125	0.752^{***}	0.227
	(0.21)	(0.19)	(0.17)	(0.17)	(0.10)	(0.10)	(0.46)	(0.38)	(0.33)	(0.30)	(0.28)	(0.26)
$ShOf_{it-1}^s$	0.605	-0.043	0.254	-0.211	0.230	0.087	0.022	-0.313^{**}	0.245^{**}	0.078	-0.225**	-0.366***
	(0.39)	(0.29)	(0.30)	(0.24)	(0.16)	(0.14)	(0.16)	(0.13)	(0.11)	(0.10)	(0.10)	(0.00)
$\Delta ShOf_{it}^{s}$	0.607	-0.302	-0.006	-0.643^{**}	0.376^{**}	0.169	0.478^{**}	-0.274^{*}	0.297^{**}	-0.077	0.067	-0.232**
	(0.44)	(0.33)	(0.34)	(0.27)	(0.18)	(0.17)	(0.20)	(0.16)	(0.14)	(0.13)	(0.12)	(0.11)
Control for output		Х		х		Х		х		Х		х
Obs.	22224	22224	22224	22224	22224	22224	18478	18478	18478	18478	18478	18478
$ShOf_{it-1}^g = \Delta ShOf_{it}^g$.001	.327	.002	.332	.412	.675	0.854	0.891	0.908	0.824	0.652	0.772
$ShOf^s_{it-1} = \Delta ShOf^s_{it}$.995	.372	.384	.078	.343	.567	0.009	0.780	0.671	0.164	0.007	0.171
Underidentification	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000
Endog. Test	0.000	.341	0.000	.118	0.000	.166	0.000	0.369	0.004	0.752	0.000	0.016
Hansen p -val.	.753	.189	.552	.151	.368	.140	0.464	0.783	0.668	0.574	0.694	0.911
$a \frac{a}{b}$ Robust standard erron ^b All specifications incluvariables, i.e. in the s, levels and growth. The Finally, the lagged lev ^c All specifications inclu	s in paren de contro pecification same holc el of empl- ide firm fiy	theses. * Is for way n where of is for cor oyment i ced effect	p < 0.10, * ges, labor p offshoring e ntrolling for s included ε	* $p < 0.05$ roductivit nters in la output. I as well.	, $* * * p < 0$ y and capi gged levels f the specif	.01 tal intens t as well s ication co	sity. The cc as in growth ontrols for e	ntrols ente 1, the cont output, ou	er in the sa rol variable tput enters	me way es also er in the se	as the offsh iter in both ame way as	oring Llagged offshoring.

TABLE 9-EMPLOYMENT GROWTH AND OFFSHORING; 1998-2005

			Manufac	turing					Ser	vices		
	(1)	(2)	(3)	(4)	(5)	(9)	(2)	(8)	(6)	(10)	(11)	(12)
	Tc	otal	Low S.	killed	High Sl	killed	Tot	al	Low S	killed	High 5	Skilled
	IV-FE	IV-FE	IV-FE	IV-FE	IV-FE	IV-FE	IV-FE	IV-FE	IV-FE	IV-FE	IV-FE	IV-FE
$ShOf_{it-1}^{g}$ OECD	0.693^{***}	0.099	0.441^{***}	0.005	0.198^{***}	0.082	1.307^{***}	0.176	0.476^{**}	-0.110	0.625^{***}	0.169
	(0.12)	(0.11)	(0.09)	(0.00)	(0.05)	(0.05)	(0.29)	(0.22)	(0.20)	(0.18)	(0.17)	(0.15)
$\Delta ShOf_{it}^g \text{ OECD}$	1.269^{***}	0.221	0.825^{***}	0.078	0.287^{***}	0.071	1.369^{***}	-0.059	0.547	-0.205	0.694^{**}	0.118
	(0.21)	(0.20)	(0.16)	(0.16)	(0.10)	(0.10)	(0.47)	(0.39)	(0.34)	(0.32)	(0.29)	(0.27)
$ShOf_{it-1}^m$ non-OECD	0.676^{**}	-0.446^{**}	0.521^{**}	-0.281	0.098	-0.134	2.850^{**}	0.925	1.937^{**}	0.973	0.512	-0.215
	(0.28)	(0.23)	(0.22)	(0.19)	(0.12)	(0.12)	(1.13)	(0.82)	(0.81)	(0.68)	(0.65)	(0.59)
$\Delta ShOf_{it}^m$ non-OECD	1.597^{***}	0.138	1.314^{***}	0.295	0.269	-0.034	6.033^{***}	2.746^{*}	4.210^{***}	2.398^{**}	1.050	-0.133
	(0.58)	(0.45)	(0.45)	(0.39)	(0.23)	(0.21)	(1.94)	(1.44)	(1.34)	(1.14)	(1.15)	(1.04)
$ShOf_{it-1}^s$ OECD	0.504	-0.028	0.229	-0.186	0.216	0.105	0.044	-0.208	0.231^{**}	0.102	-0.172^{*}	-0.277***
	(0.41)	(0.31)	(0.31)	(0.25)	(0.16)	(0.15)	(0.17)	(0.13)	(0.11)	(0.10)	(0.10)	(0.09)
$\Delta ShOf_{it}^s$ OECD	0.692	-0.164	0.044	-0.577**	0.419^{**}	0.238	0.367^{*}	-0.286^{*}	0.222	-0.097	0.046	-0.217^{**}
	(0.44)	(0.33)	(0.34)	(0.28)	(0.18)	(0.17)	(0.19)	(0.15)	(0.14)	(0.12)	(0.12)	(0.11)
$ShOf_{it-1}^{s}$ non-OECD	-0.020	1.124	2.222	2.633	-1.431	-1.126	3.412^{**}	0.466	1.566	0.131	1.308	0.204
	(3.10)	(2.25)	(2.33)	(1.87)	(1.29)	(1.17)	(1.56)	(1.22)	(1.04)	(0.92)	(0.99)	(0.89)
$\Delta ShOf_{it}^s$ non-OECD	1.648	2.464	3.779	4.080^{*}	-1.132	-0.878	5.392^{**}	-0.448	2.670^{*}	-0.276	2.633^{*}	0.443
	(3.69)	(2.72)	(2.70)	(2.18)	(1.54)	(1.38)	(2.31)	(1.90)	(1.53)	(1.41)	(1.52)	(1.40)
Control for output		Х		Х		Х		х		Х		х
Obs	24498	24498	24498	24498	24498	24498	20177	20177	20177	20177	20177	20177
Underidentification	.000	.000	.000	000.	000.	000.	000.	.000	000	000.	000.	.000
Endog. Test	000.	.421	.000	.314	000	.367	000	.433	000	.578	.005	.468
Hansen p -val	.045	.011	.085	.013	.223	.179	.188	.325	.250	.114	.578	.937
$\frac{a}{b}$ Robust standard err ^b All specifications in offshoring variables, in both lagged level same way as offshor	cors in par clude cont i.e. in the s and grov ing. Final	rentheses. Trols for we e specifica wth. The se ly, the lag	p < 0.10, ages, labor tion where une holds f ged level o	$\begin{array}{l} **p < 0.\\ \text{producti}\\ \text{offshorin}\\ \text{for contro}\\ f \text{ employr} \end{array}$	05, * * *p. vity and c g enters in fling for ou	< 0.01 apital in a lagged utput. If	tensity. Tl levels as w the speci well.	he contrc rell as in fication c	ols enter ir growth, tl controls fo	the sam he contro r output,	e way as tl l variables output en	ae also enter cers in the
^c All specifications in	clude firm	fixed effe	cts.									

TABLE 10-EMPLOYMENT GROWTH AND OFFSHORING; GOODS AND SERVICES SEPARATELY; OECD VERSUS NON-OECD

Appendices

A. Classification of Services

Table A.1 lists the types of services and the corresponding NACE code. The table further aggregates services into seven categories: Transport, Financial and Insurance, Construction, IT Business and Cultural Services.

Rub	EBOPS	Description of service in services trade database	NACE 1.1	EU KLEMS	Category
100 101 102	207 208 209	Transportation - Sea transport - Passenger Transportation - Sea transport - Freight Transportation - Sea transport - Other	61	60t63	
110 111 112	211 212 213	Transportation - Air transport - Passenger Transportation - Air transport - Freight Transportation - Air transport - Other	62	60t63	Transport
120 121 122	215 216 217	Transportation - Other transport - Passenger Transportation - Other transport - Freight Transportation - Other transport - Other	60 + 63	60t63	
130 131	260 260	Financial services Financial services	65t67	J	Financial/Insurance
142 143 144	246 246 247	Communication services - Postal and courier services Communication services - Postal and courier services Communication services - Telecommunication services	64	64	Communication
156	249	Construction Services	45	F	Construction
162 163	$263 \\ 264$	Computer and information services - Computer services Computer and information services - Information services	72	71t74	IT
180 181 182 183 184	$271 \\ 256 \\ 254 \\ 255 \\ 257$	Insurance services - Other direct insurance Insurance services - Life insurance and pension funding Insurance services - Freight insurance Insurance services - Reinsurance Insurance services - Auxiliary services	65t67	J	Financial/Insurance
190	258	Other business services - Research and development	73	71t74	
191 193	274 280	Other business services - Miscellaneous business, professional and technical services - Legal, accounting, management, consulting and public relations Other business services - Miscellaneous business, professional and technical services - Architectural, engineering and other technical	74	71t74	Business services
194	278	Services Other business services - Miscellaneous business, professional and technical services - Advertising, market research, and public opin- ion polling			
195	283	Other business services - Miscellaneous business, professional and technical services - Agricultural, mining and on-site processing services - Other			
196	282	Other business ervices - Miscellaneous business, professional and technical services - Agricultural, mining and on-site processing corridec. Worth tractment and depallution			
199	284	Services - waste treatment and deponution Other business services - Miscellaneous business, professional and technical services - Other			
231 239	288 289	Personal, cultural and recreational services: Audiovisual and re- lated services Personal, cultural and recreational services: Other personal, cul- tural and recreational activities	92	0	Cultural
247		DROPPED from data, does not exist in classification	-		
270	272	Other business services - Operational leasing services	71	71t74	Business services

TABLE A.1—INTERNATIONAL TRADE OF SERVICES: CLASSIFICATION OF SERVICES

^a Column "Rub" lists the services codes appearing in the Belgian data. The column "EBOPS" lists the corresponding (international) EBOPS code for that service. The column "NACE Rev 1.1" lists the corresponding NACE sector for each service, while the column "EU KLEMS" assigns a NACE sector code that features in the EU KLEMS data (sometimes these sectors are more aggregated than the original NACE codes. The final column "Service category" shows the groupings applied in the data. We consider these services categories separately in the data. All services except "Transport" are "producer services", we have already excluded e.g. travel services from the data prior to assigning service categories.

B. Summaries Evolution Trade in Goods and Services

Table B.1 displays services imports and exports for the years 1995 and 2005. The total value of service imports surged from 8.5 billion euros in 1995 to 20 billion euros in 2005, an increase of over 130%. Service exports increased by 145%, from 10 billion euros to 24 billion euros. Transport services are the largest category, followed by Business Services and Financial Services. The rankings of the different service types are similar for imports and exports, consistent with models of intra-industry trade and product differentiation. Growth in Transport Services follows the trend in goods trade more closely and decreases in relative importance over the sample period. International trade in Business, ICT and Communication Services boomed particularly between 1995 and 2005.

The last row in Table B.1 displays total imports and exports of goods in 1995 and 2005. While the increase in Service Trade was more pronounced over the period, the value of total trade in services was only 10% of the value of total trade in goods in 2005. Table B.2 makes a comparison between the top destination countries/countries of origin for exports and imports of goods versus services. Trade in goods is dominated by trade with the three neighbouring countries (The Netherlands, Germany and France), accounting for over 40% of total trade in goods. For trade in services, the United Kingdom tops the ranking of the countries of destination/origin and the US ranks second for exports. This is consistent with the findings of Breinlich and Criscuolo (2011) that the US and the UK are more important trading partners for services than for goods trade.

			Imports					Exports		
	19	995	200)5	Change	19	95	20	05	Change
	Value	Share	Value	Share	95-05	Value	Share	Value	Share	95-05
Transport Services	4,628	53.4%	8,802	44.1%	+90.2%	5,639	56.5%	9,922	40.7%	+75.9%
Business Services	1,604	18.5%	5,692	28.5%	+254.8%	1,645	16.5%	6,214	25.5%	+277.8%
Financial/Insurance Services	1,112	12.8%	1,929	9.7%	+73.4%	1,122	11.2%	2,800	11.5%	+149.5%
Construction Services	442	5.1%	636	3.2%	+43.8%	410	4.1%	1,439	5.9%	+251.3%
IT Services	398	4.6%	1,476	7.4%	+271.2%	656	6.6%	1,975	8.1%	+201.0%
Communication Services	238	2.7%	1,048	5.3%	+340.8%	418	4.2%	1,644	6.7%	+293.3%
Cultural Services	237	2.7%	370	1.9%	+55.7%	86	0.9%	411	1.7%	+376.2%
Total Trade in Serices	8,659	100.0%	19,952	100%	+130.4%	9,977	100.0%	24,406	100.0%	+144.6%
Trade in Goods	99,740		214,540		+115.0%	110,352		224,976		+103.9%
VI NOUS D DU	1 6 5 1	12 14	1.1	1.1	1.	1	1 4 1			

TABLE B.1—INTERNATIONAL TRADE IN GOODS AND SERVICES, 1995-2005

Services					
Imports			Exports		
Top Countries	Value	Share	Top Countries	Value	Share
United Kingdom	3,226	16.2%	United Kingdom	4,260	17.5%
France	2,502	12.5%	USA	4,147	17.0%
The Netherlands	2,479	12.4%	The Netherlands	$3,\!600$	14.7%
USA	2,311	11.6%	France	2,888	11.8%
Germany	2,212	11.1%	Germany	2,706	11.1%
Italy	870	4.4%	Luxembourg	1,136	4.7%
Switzerland	726	3.6%	Switzerland	730	3.0%
Spain	574	2.9%	Sweden	487	2.0%
Luxembourg	522	2.6%	Italy	455	1.9%
Hong Kong	403	2.0%	Spain	351	1.4%
Goods					
Imports			Exports		
Top Countries	Value	Share	Top Countries	Value	Share
The Netherlands	51,272	20.5%	Germany	42,383	16.1%
Germany	35,918	14.3%	France	40,669	15.5%
France	27,457	11.0%	The Netherlands	32,798	12.5%
USA	14,362	5.7%	United Kingdom	19,186	7.3%
United Kingdom	14,356	5.7%	USA	15,599	5.9%
Ireland	14,254	5.7%	Italy	12,107	4.6%
China	8,035	3.2%	Spain	8,313	3.2%
Italy	6,987	2.8%	India	6,661	2.5%
Japan	5,942	2.4%	Luxembourg	4,973	1.8%
Russia	5.935	2.4%	Poland	4.479	1.7%

TABLE B.2—Countries of Origin/Destination and Trade in Services/Goods

 Russia
 3,955
 2.4%
 Foland
 4,479
 1.7

 Values in Million Euros. Services data are from 2005. Goods data are from 2010

C. Fixed Effects Results

	To	otal	Low S	killed	High	Skilled
	(1)	(2)	(3)	(4)	(5)	(6)
	\mathbf{FE}	\mathbf{FE}	\mathbf{FE}	\mathbf{FE}	\mathbf{FE}	\mathbf{FE}
$ShOf_{it-1}^{g}$	0.111^{***}	-0.033	0.088^{***}	-0.016	0.022^{**}	-0.009
	(0.02)	(0.02)	(0.02)	(0.02)	(0.01)	(0.01)
$\Delta ShOf_{it}^{g}$	0.084^{***}	-0.069***	0.074^{***}	-0.034^{*}	0.007	-0.028***
	(0.02)	(0.02)	(0.02)	(0.02)	(0.01)	(0.01)
$ShOf_{it-1}^s$	0.070	0.050	0.118	0.103	-0.074	-0.078
	(0.12)	(0.11)	(0.09)	(0.08)	(0.05)	(0.05)
$\Delta ShOf_{it}^s$	0.025	0.020	-0.023	-0.027	0.023	0.023
	(0.09)	(0.07)	(0.07)	(0.06)	(0.04)	(0.04)
Controls for Output		Х		Х		Х
Obs	24489	24489	24489	24489	24489	24489
$ShOf_{it-1}^g = \Delta ShOf_{it}^g$.184	.060	.368	.241	.121	.044
$ShOf_{it-1}^s = \Delta ShOf_{it}^s$.634	.747	.051	.047	.028	.028

TABLE C.1—EMPLOYMENT GROWTH AND OFFSHORING IN MANUFACTURING SECTORS; LOW VERSUS HIGH EDUCATED; FIXED EFFECTS

 $\overline{{}^a$ Robust standard errors in parentheses. *p < 0.10, **p < 0.05, ***p < 0.01

 b All specifications include controls for wages, labor productivity and capital intensity. The controls enter in the same way as the offshoring variables, i.e. in the specification where offshoring enters in lagged levels as well as in growth, the control

variables also enter in both lagged levels and growth. The same holds for controlling for output. If the specification controls for output, output enters in the same way as offshoring. Finally, the lagged level of employment is included as well.

^c All specifications include firm fixed effects.

		Total	Lo	w Skilled	H	ligh Skilled
	(1)	(2)	(3)	(4)	(5)	(6)
	\mathbf{FE}	\mathbf{FE}	FE	\mathbf{FE}	FE	\mathbf{FE}
$ShOf_{it-1}^{g}$	0.134^{*}	-0.038	0.111**	0.023	0.026	-0.042
	(0.07)	(0.07)	(0.05)	(0.04)	(0.03)	(0.03)
$\Delta ShOf_{it}^{g}$	0.188^{***}	-0.032	0.106**	-0.007	0.092**	0.005
	(0.07)	(0.06)	(0.05)	(0.05)	(0.04)	(0.04)
$ShOf_{it-1}^s$	-0.100**	-0.098***	-0.015	-0.015	-0.082***	-0.082***
	(0.05)	(0.04)	(0.03)	(0.03)	(0.03)	(0.03)
$\Delta ShOf_{it}^s$	-0.120***	-0.144***	-0.056*	-0.068**	-0.062**	-0.071***
	(0.04)	(0.04)	(0.03)	(0.03)	(0.03)	(0.02)
Controls for Output		Χ		X		Χ
Obs	20161	20161	20161	20161	20161	20161
$ShOf_{it-1}^g = \Delta ShOf_{it}^g$.329	.900	.907	.407	.041	.139
$ShOf_{it-1}^s = \Delta ShOf_{it}^s$.633	.207	.173	.056	.442	.670

TABLE C.2—Employment growth and offshoring in services sectors; Low versus high EDUCATED; FIXED EFFECTS

 $\overline{{}^a$ Robust standard errors in parentheses. *p < 0.10, **p < 0.05, **p < 0.01 b All specifications include controls for wages, labor productivity and capital intensity. The controls enter in the same way as the offshoring variables, i.e. in the specification where offshoring enters in lagged levels as well as in growth, the control variables also enter in both lagged levels and growth. The same holds for controlling for output. If the specification controls for output, output enters in the same way as offshoring. Finally, the lagged level of employment is included as well.

^c All specifications include firm fixed effects.

]	Total	Low	Skilled	High	n Skilled
	(1)	(2)	(3)	(4)	(5)	(6)
	\mathbf{FE}	\mathbf{FE}	\mathbf{FE}	\mathbf{FE}	FE	\mathbf{FE}
$ShOf_{it-1}^{g}$	0.062^{*}	-0.077***	0.048**	-0.053**	0.016	-0.013
	(0.03)	(0.03)	(0.02)	(0.02)	(0.01)	(0.01)
$\Delta ShOf_{it}^{g}$	0.037	-0.152^{***}	0.053^{*}	-0.080***	-0.008	-0.051^{***}
	(0.04)	(0.03)	(0.03)	(0.03)	(0.01)	(0.01)
$ShOf_{it-1}^s$	-0.042	0.086	0.069	0.160^{*}	-0.136	-0.107
	(0.16)	(0.15)	(0.10)	(0.09)	(0.09)	(0.09)
$\Delta ShOf_{it}^s$	0.078	0.162	-0.002	0.056	0.024	0.043
	(0.13)	(0.11)	(0.09)	(0.08)	(0.06)	(0.06)
Controls for Output		Х		Х		Х
Obs	24518	24518	24518	24518	24518	24518
$ShOf_{it-1}^g = \Delta ShOf_{it}^g$.445	.010	.854	.244	.106	.007
$ShOf_{it-1}^s = \Delta ShOf_{it}^s$.308	.551	.407	.225	.022	.038

TABLE C.3—EMPLOYMENT GROWTH AND OFFSHORING IN MANUFACTURING SECTORS; NAR-ROW OFFSHORING OF GOODS AND SERVICES OFFSHORING EXCLUDES TRANSPORT SER-VICES; FIXED EFFECTS

 $\overline{^a$ Robust standard errors in parentheses. *p < 0.10, **p < 0.05, ***p < 0.01

^b All specifications include controls for wages, labor productivity and capital intensity. The controls enter in the same way as the offshoring variables, i.e. in the specification where offshoring enters in lagged levels as well as in growth, the control variables also enter in both lagged levels and growth. The same holds for controlling for output. If the specification controls for output, output enters in the same way as offshoring. Finally, the lagged level of employment is included as well. ^c All specifications include firm fixed effects.

	To	tal	Low S	Skilled	High	Skilled
	(1)	(2)	(3)	(4)	(5)	(6)
	\mathbf{FE}	\mathbf{FE}	FE	\mathbf{FE}	FE	\mathbf{FE}
$ShOf_{it-1}^{g}$	0.121^{*}	-0.039	0.105**	0.023	0.021	-0.042
	(0.07)	(0.06)	(0.05)	(0.04)	(0.03)	(0.03)
$\Delta ShOf_{it}^{g}$	0.184^{***}	-0.035	0.105**	-0.007	0.088**	0.002
	(0.07)	(0.06)	(0.05)	(0.04)	(0.04)	(0.04)
$ShOf_{it-1}^s$ Core	-0.110	-0.127**	0.005	-0.004	-0.099**	-0.106^{***}
	(0.07)	(0.05)	(0.04)	(0.04)	(0.04)	(0.04)
$\Delta ShOf_{it}^s$ Core	-0.072	-0.152^{**}	-0.062	-0.103**	-0.017	-0.049
	(0.07)	(0.06)	(0.05)	(0.05)	(0.05)	(0.04)
Controls for Output		X		X		X
Obs	20275	20275	20275	20275	20275	20275
$ShOf_{it-1}^g = \Delta ShOf_{it}^g$.262	.928	.999	.405	.033	.154
$ShOf_{it-1}^s = \Delta ShOf_{it}^s$.586	.664	.170	.028	.069	.166

TABLE C.4—Employment growth and offshoring in services sectors; Core Services; Fixed Effects

 a Robust standard errors in parentheses. *p < 0.10, **p < 0.05, ***p < 0.01

^b All specifications include controls for wages, labor productivity and capital intensity. The controls enter in the same way as the offshoring variables, i.e. in the specification where offshoring enters in lagged levels as well as in growth, the control variables also enter in both lagged levels and growth. The same holds for controlling for output. If the specification controls for output, output enters in the same way as offshoring. Finally, the lagged level of employment is included as well.

^c All specifications include firm fixed effects.

			Manufa	cturing					Serv	ices		
	(1)	(2)	(3)	(4)	(5)	(9)	(1)	(2)	(2)	(8)	(13)	(14)
	To	tal	Low 5	Skilled	High	Skilled	To	tal	Low S	killed	High S	skilled
	FE	FE	FE	FE	FE	FE	FE	FE	FE	FE	FE	FE
$ShOf_{it-1}^{g}$ OECD	0.130^{***}	-0.019	0.104^{***}	-0.003	0.023^{**}	-0.009	0.111	-0.057	0.099^{**}	0.013	0.018	-0.049
	(0.03)	(0.02)	(0.02)	(0.02)	(0.01)	(0.01)	(0.08)	(0.01)	(0.05)	(0.05)	(0.03)	(0.03)
$\Delta ShOf_{it}^g$ OECD	0.104^{***}	-0.055^{**}	0.090^{***}	-0.021	0.008	-0.028***	0.159^{**}	-0.074	0.074	-0.045	0.099^{**}	0.006
	(0.03)	(0.02)	(0.02)	(0.02)	(0.01)	(0.01)	(0.01)	(0.01)	(0.05)	(0.05)	(0.04)	(0.04)
$ShOf_{it-1}^m$ non-OECD	-0.039	-0.163^{***}	-0.023	-0.113^{**}	0.002	-0.024	0.434^{**}	0.126	0.221	0.063	0.148	0.025
1	(0.01)	(0.05)	(0.06)	(0.05)	(0.02)	(0.02)	(0.22)	(0.18)	(0.18)	(0.13)	(0.12)	(0.09)
$\Delta ShOf_{it}^m \text{ non-OECD}$	-0.102	-0.262^{***}	-0.075	-0.188^{***}	0.008	-0.027	0.922^{**}	0.605^{*}	0.734^{***}	0.572^{**}	0.188	0.062
	(0.00)	(0.07)	(0.07)	(0.06)	(0.04)	(0.03)	(0.38)	(0.32)	(0.28)	(0.25)	(0.21)	(0.19)
$ShOf_{it-1}^s$ OECD	0.035	0.021	0.097	0.087	-0.084	-0.087	-0.122**	-0.110^{***}	-0.031	-0.025	-0.089***	-0.084***
1	(0.13)	(0.13)	(0.09)	(0.08)	(0.05)	(0.06)	(0.05)	(0.04)	(0.03)	(0.03)	(0.03)	(0.03)
$\Delta ShOf_{it}^s$ OECD	-0.004	-0.017	-0.036	-0.045	0.014	0.011	-0.136^{***}	-0.146^{***}	-0.072^{**}	-0.077**	-0.068**	-0.073***
	(0.00)	(0.08)	(0.07)	(0.06)	(0.04)	(0.04)	(0.04)	(0.04)	(0.03)	(0.03)	(0.03)	(0.03)
$ShOf_{it-1}^s$ non-OECD	1.272	0.165	1.306	0.500	0.006	-0.222	0.743	0.318	0.802	0.585	-0.210	-0.379
1	(1.11)	(0.99)	(0.93)	(0.84)	(0.57)	(0.56)	(0.85)	(0.61)	(0.50)	(0.42)	(0.51)	(0.42)
$\Delta ShOf_{it}^s$ non-OECD	1.633	1.528	1.549	1.440	0.237	0.239	1.992^{**}	0.917	1.235^{**}	0.688	0.757	0.331
	(1.35)	(1.28)	(1.00)	(0.93)	(0.76)	(0.76)	(0.88)	(0.77)	(0.57)	(0.55)	(0.56)	(0.53)
Control for output		Х		Х		х		Х		Х		Х
Obs	24498	24498	24498	24498	24498	24498	20177	20177	20177	20177	20177	20177
$\frac{a}{b}$ Robust standard eri ^b All specifications in	cors in par clude contr	entheses. * rols for wag	p < 0.10, * ges, labor]	productivit	, * * * p < y and cap	0.01 bital intens	ity. The cc	ntrols ente	r in the sa	me way a	s the offsho	bring Lowed
levels and growth.T.	he same ho	olds for con	trolling for	r output. I	f the spec	ification co	ontrols for e	output, out	put enters	in the sa	me way as	offshoring.
Finally, the lagged 1 ^c All specifications inc	evel of em	ployment i fixed effect	s included s.	as well.							5)

TABLE C.5-EMPLOYMENT GROWTH AND OFFSHORING; GOODS AND SERVICES SEPARATELY; OECD VERSUS NON-OECD

49

NATIONAL BANK OF BELGIUM - WORKING PAPERS SERIES

The Working Papers are available on the website of the Bank: http://www.nbb.be.

- 276. "How do exporters react to changes in cost competitiveness?, by S. Decramer, C. Fuss and J. Konings, *Research series*, January 2015.
- 277. "Optimal monetary policy response to endogenous oil price fluctuations", by A. Stevens, *Research series*, January 2015.
- 278. "Comparing fiscal multipliers across models and countries in Europe", by J. Kilponen, M. Pisani, S. Schmidt, V. Corbo, T. Hledik, J. Hollmayr, S. Hurtado, P. Júlio, D. Kulikov, M. Lemoine, M. Lozej, H. Lundvall, J. R. Maria, B. Micallef, D. Papageorgiou, J. Rysanek, D. Sideris, C. Thomas and G. de Walque, *Research series*, March 2015.
- 279. "Assessing European competitiveness: The new CompNet micro-based database", by P. Lopez-Garcia, F. di Mauro and the CompNet Task Force, *Research series*, April 2015.
- 280. "FloGARCH: Realizing long memory and asymmetries in returns volatility", by H. Vander Elst, *Research series*, April 2015.
- 281. "Does education raise productivity and wages equally? The moderating roles of age, gender and industry", by F. Rycx, Y. Saks and I. Tojerow, *Research series,* April 2015.
- 282. "Assessing European firms' exports and productivity distributions: The CompNet trade module", by A. Berthou, E. Dhyne, M. Bugamelli, A.-M. Cazacu, C.-V. Demian, P. Harasztosi, T. Lalinsky, J. Merikül, F. Oropallo and A. C. Soares, *Research series*, May 2015.
- 283. "Economic importance of the Belgian ports: Flemish maritime ports, Liège port complex and the port of Brussels Report 2013", by F. Van Nieuwenhove, *Document series*, June 2015.
- 284. "Crisis-proof services: Why trade in services did not suffer during the 2008-2009 collapse", by A. Ariu, *Research series*, July 2015.
- 285. "The labour market position of second-generation immigrants in Belgium", by V. Corluy, J. Haemels, I. Marx and G. Verbist, *Research series*, September 2015.
- 286. "The implications of household size and children for life-cycle saving", by B. Capéau and B. De Rock, *Research series*, September 2015.
- 287. "Monetary policy effects on bank risk taking", by A. Abbate and D. Thaler, *Research series*, September 2015.
- 288. "The Belgian production network 2002-2012", by E. Dhyne, G. Magerman and S. Rubínová, *Research series*, October 2015.
- 289. "Portfolio choice and investor preferences: A semi-parametric approach based on risk horizon", by G. Hübner and T. Lejeune, *Research series*, October 2015.
- 290. "Predicting Belgium's GDP using targeted bridge models", by Ch. Piette, Research series, January 2016.
- 291. "Did export promotion help firms weather the crisis?", by J. Van Biesebroeck, J. Konings and C. Volpe Martineus, *Research series*, January 2016.
- 292. "On the role of public policies and wage formation for business investment in R&D: A long-run panel analysis", by T. Buyse, F. Heylen and R. Schoonackers, *Research series*, January 2016.
- 293. "Unraveling firms: Demand, productivity and markups heterogeneity", by E. Forlani, R. Martin, G. Mion and M. Muûls, *Research series*, February 2016.
- 294. "Unemployment risk and over-indebtedness: A micro-econometric perspective", by Ph. Du Caju, F. Rycx and I. Tojerow, *Research series*, February 2016.
- 295. "International shocks and domestic prices: How large are strategic complementarities?", by A. Amiti, O. Itskhoki and J. Konings, *Research series*, March 2016.
- 296. "The supplier network of exporters: Connecting the dots", by E. Dhyne and S. Rubínová, *Research series*, May 2016.
- 297. "Does one size fit all at all times? The role of country specificities and state dependencies in predicting banking crises" by S. Ferrari and M. Pirovano, *Research series,* May 2016.
- 298. "Competition and product mix adjustment of multi-product exporters: Evidence from Belgium", by K. Breemersch, *Research series*, June 2016.
- 299. "Flemish maritime ports, Liège port complex and the port of Brussels Report 2014", by G. Van Gastel, *Document series*, June 2016.
- 300. "Misalignment of productivity and wages across regions? Evidence from Belgian matched panel data", by F. Rycx, Y. Saks and I. Tojerow, *Research series*, July 2016.
- 301. "The European Payments Union and the origins of Triffin's regional approach towards international monetary integration", by I. Maes and I. Pasotti, *Research series,* September 2016.
- 302. "The transmission mechanism of credit support policies in the Euro Area", by J. Boeckx, M. de Sola Perea and G. Peersman, *Research series*, October 2016.

- 303. "Bank capital (requirements) and credit supply: Evidence from pillar 2 decisions", by O. De Jonghe, H. Dewachter and S. Ongena, *Research series*, October 2016.
- 304. "Monetary and macroprudential policy games in a monetary union", by R. Dennis and P. Ilbas, *Research series*, October 2016.
- 305. "Forward guidance, quantitative easing, or both?, by F. De Graeve and K. Theodoridis, *Research series*, October 2016.
- 306. "The impact of sectoral macroprudential capital requirements on mortgage loan pricing: Evidence from the Belgian risk weight add-on", by S. Ferrari, M. Pirovano and P. Rovira Kaltwasser, *Research series*, October 2016.
- 307. "Assessing the role of interbank network structure in business and financial cycle analysis", by J-Y Gnabo and N.K. Scholtes, *Research series*, October 2016.
- 308. "The trade-off between monetary policy and bank stability", by M. Lamers, F. Mergaerts, E. Meuleman and R. Vander Vennet, *Research series*, October 2016.
- 309. "The response of euro area sovereign spreads to the ECB unconventional monetary policies", by H. Dewachter, L. Iania and J-C. Wijnandts, *Research series*, October 2016.
- 310. "The interdependence of monetary and macroprudential policy under the zero lower bound", by V. Lewis and S. Villa, *Research series*, October 2016.
- 311. "The impact of exporting on SME capital structure and debt maturity choices", by E. Maes, N. Dewaelheynes, C. Fuss and C. Van Hulle, *Research series*, October 2016.
- 312. "Heterogeneous firms and the micro origins of aggregate fluctuations", by G. Magerman, K. De Bruyne, E. Dhyne and J. Van Hove, *Research series*, October 2016.
- 313. "A dynamic factor model for forecasting house prices in Belgium", by M. Emiris, *Research series*, November 2016.
- 314. "La Belgique et l'Europe dans la tourmente monétaire des années 1970 Entretiens avec Jacques van Ypersele", by I. Maes and S. Péters, *Research series*, December 2016.
- 315. "Creating associations to substitute banks' direct credit. Evidence from Belgium", by M. Bedayo, *Research series*, December 2016.
- 316. "The impact of export promotion on export market entry", by A. Schminke and J. Van Biesebroeck, *Research series*, December 2016.
- 317. "An estimated two-country EA-US model with limited exchange rate pass-through", by G. de Walque, Ph. Jeanfils, T. Lejeune, Y. Rychalovska and R. Wouters, *Research series,* March 2017.
- 318. Using bank loans as collateral in Europe: The role of liquidity and funding purposes", by F. Koulischer and P. Van Roy, *Research series*, April 2017.
- 319. "The impact of service and goods offshoring on employment: Firm-level evidence", by C. Ornaghi, I. Van Beveren and S. Vanormelingen, *Research series*, May 2017.

National Bank of Belgium Limited liability company RLP Brussels – Company's number: 0203.201.340 Registered office: boulevard de Berlaimont 14 – BE-1000 Brussels www.nbb.be

Editor

Jan Smets Governor of the National Bank of Belgium

© Illustrations: National Bank of Belgium

Layout: Analysis and Research Group Cover: NBB AG – Prepress & Image

Published in May 2017